

Working Paper 47/06 – Rev. 2011

ACTUARIAL FAIRNESS WHEN LONGEVITY INCREASES: AN EVALUATION OF THE ITALIAN PENSION SYSTEM

Michele Belloni Carlo Maccheroi

Actuarial fairness when longevity increases: an evaluation of the Italian pension system

MICHELE BELLONI * †

University of Turin, CeRP-Collegio Carlo Alberto and NETSPAR (*e-mail:* belloni@cerp.unito.it)

CARLO MACCHERONI

University of Turin and University of Milan "L. Bocconi" (e-mail: cmaccheroni@econ.unito.it)

Abstract

In this study, we analyse the actuarial features of the Italian pension system after the 1995 reform. We consider both the old defined benefit, the pro-rata and the new notional defined contribution pension rules applied to private sector employees born between 1945 and 2000. In the computations, we allow for dynamic mortality. To this aim, we project cohort- and gender-specific mortality rates based on a limit demographic scenario recently depicted by demographic experts. We compare findings for the current legislation with those from a quasi-actuarially fair scenario, where cohort- and gender-specific mortality rates are taken into account in the pension computation.

The old DB rules are extremely generous and offer strong incentives to early retirement. Due to dynamic efficiency, the new NDC scheme provides less than actuarially fair benefits. As a consequence of the rules adopted to compute coefficients used to convert the notionally accumulated sum at retirement into the annuity and to update them in response to increasing longevity, the NDC scheme is more than quasiactuarially fair. Periodical *ex-post* adjustments of the coefficients generate big incentives to retire. The main cause for the actuarial unfairness embedded in the new Italian pension system is the use of crosssectional mortality rates in the computation of conversion coefficients: retired cohorts will likely live longer than what accounted for in the computation of their pension benefits, since cohort effects in mortality are disregarded by the Italian law.

Keywords: social security, notional defined contribution pension systems, actuarial fairness, longevity, cohort-specific mortality forecasts. JEL codes: H55, J11, J14.

^{*} Address for correspondence: Collegio Carlo Alberto, Via Real Collegio 30, 10024 Moncalieri (TO), Italy.

[†] The authors thank E. Fornero, the participants to the seminars at Tinbergen Institute Amsterdam and CeRP-Collegio Carlo Alberto, to the workshop for the presentation of Ph.D. DSE thesis of the XVII cycle at the University of Turin, to the 2003 annual conference of the International Social Security Administration in Antwerp and to the annual meetings of the COFIN research group for their useful comments. The research was supported by a grant from MIUR (n. 2002138899 003 "Metodi e strumenti per l'analisi e la gestione dei rischi nei settori delle assicurazione di persone e previdenziale"). The opinions expressed herein are those of the authors. All mistakes and/or errors are entirely the authors' responsibility.

1 Introduction

In the last two decades, population aging and low economic growth have undermined the financial stability of several pay-as-you-go pension schemes. Some countries (including Italy, Latvia, Poland and Sweden) have reacted to the crises by replacing their defined benefit (DB) pension system with a notional defined contribution (NDC) system, keeping the previous pay-as-you-go financial architecture. An NDC scheme consists of an individual account system to which contributions are earmarked and interests notionally paid; at retirement, the notionally accumulated sum is converted into the pension taking into account life expectancy, i.e. incorporating actuarial fairness.

An ideal actuarially fair pension scheme is characterized by both actuarial fairness and actuarial fairness at margin. Actuarial fairness guarantees that, for each individual, the discounted sum of contributions paid during the working career is equal to the discounted sum of expected benefits. On the other hand, actuarial fairness at margin ensures that, for each individual, the discounted sum of expected benefits (net of expected contributions to pay in case of continued work) does not depend on the worker's retirement age (see e.g Legros 2006, Gruber and Wise 1999).

An NDC scheme cannot guarantee the two above described "pure" actuarial conditions; however, it must satisfy the weaker conditions of *quasi*-actuarial fairness on average, and *quasi*-actuarial fairness at margin on average (Palmer 2006). These conditions first recognize that, in a dynamic efficient economy - i.e. an economy where the risk-adjusted return on assets is greater than the earnings growth rate - only a funded DC pension scheme can guarantee pure actuarial fairness. Since an NDC system is pay-as-you-go financed, its feasible (equilibrium) rate of return is approximately equal to the wage bill growth rate, i.e. the sum of earnings growth per head and the population growth (see Aaron 1966, Samuelson 1958). Moreover, these conditions allow for actuarial fairness (at margin) to only hold "on average", whereas in a pure actuarial system they must hold for each individual.¹

The implementation of a *quasi*-actuarially fair NDC system needs to handle increasing longevity, and to insure that life expectancy used to compute the annuity at retirement is as close as possible to actual, *ex-post*, residual life (Disney 2004).² Palmer (2006) has recently suggested three procedures to handle increasing longevity in NDC schemes. The first procedure consists in forming a committee of demographic experts in charge of the analysis of long-run scenarios and the publication of official cohort projections. Revisions would always apply to non-retired cohorts and would be increasingly small as the cohort approaches retirement age. The second procedure consists in estimating cohort-specific mortality tables based on historical cross-sectional survival data. In this case, periodical adjustments are necessary to avoid producing excessively generous, i.e. system-

¹ NDC systems should account for known differences in life expectancy related to individual characteristics such as gender, race (Sorlie, Rogot, Anderson, Johnson, and Backlund 1992), socioeconomic status (Kitagawa and Hauser 1973) and region (Caselli, Peracchi, Balbi, and Lipsi 2003).

 $^{^2}$ DG-ECFIN (2006) projections state that life expectancy at age 65 for the EU-25 countries will increase by about 4 years by 2050.

atically more-than-actuarially fair, annuities. The third procedure makes use of recent cross-sectional survival data to compute annuities at retirement; it regularly adjusts the pensions of all retirees on the basis of more updated statistics. According to Palmer (2006), the first and the third procedures have the advantage of explicitly fulfilling the criterion of financial equilibrium, whereas the second and the third are, in principle, less discretionary than the first since they are based on historical statistics.

In 1995, the Italian pension system was reformed, and an NDC scheme was introduced to replace the previous unsustainable DB scheme. The reform set up a transitional phase lasting almost thirty years toward the new rules. Therefore, in the coming two decades, pensions will still be computed according to the old rules (the so-called *pro-rata* system); consequently, their actuarial features will still depend on the characteristics of the old DB scheme. New DC pensions, on the other hand, will be computed on the basis of updated cross-sectional mortality tables, while benefits of retired individuals will be kept untouched regardless of longevity changes. From a comparison with the third procedure suggested by Palmer (2006), it is clear that the Italian NDC rules will systematically violate actuarial principles if longevity continues to rise.

Various studies have analysed the actuarial characteristics of the Italian pension system after the 1995 reform. Ferraresi and Fornero (2000) and Fornero and Castellino (2001) analyse a set of workers which is representative of future cohorts of retirees. Relying on a static mortality assumption, these scholars provide empirical evidence suggesting that the DB scheme is extremely generous, whereas the NDC is almost actuarially fair on average and at margin.³ While Ferraresi and Fornero (2000) and Fornero and Castellino (2001) focus on differences across cohorts, others (e.g. Caselli, Peracchi, Balbi, and Lipsi (2003) find evidence of a sizable redistribution generated by the DC scheme which occurs across genders and Italian regions and which is induced by differential mortality. Borella and Coda (2006) develop a microsimulation model to study the redistributive impact of the Italian pension system both between and within cohorts; simulations are all based on cross-sectional mortality tables (ISTAT 2000).

In this study, we follow the representative agents approach proposed by Ferraresi and Fornero (2000) and Fornero and Castellino (2001). We analyse the actuarial features of the Italian pension system after the 1995 reform. We consider both the DB, the *pro-rata* and the (N)DC steady state rules applied to INPS-FPLD (i.e. private sector) employees. We compute two actuarial indicators, namely the present value ratio (PVR) and the implicit tax rate (TAX). Differently from previous studies, we allow for dynamic mortality in their computation. In particular, we investigate the actuarial properties of the rules designed by the 1995 reform to handle increasing longevity. To this aim, we build projected cohort- and gender-specific mortality tables based on a limit demographic scenario recently depicted by Robine, Crimmins, Horiuchi, and Zeng (2006). We first use the projected life tables in a baseline scenario (scenarios "B95" and "B07", see section 4) to simulate pension benefits

³ These results are qualitatively confirmed by other studies (see e.g. Brugiavini and Peracchi 2004, Brugiavini and Peracchi 2003) which quantified the degree of actuarial fairness at margin of the Italian pension system in order to gauge its effects on retirement choices.

according to the current legislation. We then compare these results with alternative findings from a *quasi*-actuarial fair scenario ("AB") where conversion coefficients are cohort-(and gender-)specific, as indicated by the first procedure in Palmer (2006). Although our analysis focuses on actuarial differences across cohorts, in section 7, we also provide some evidence of within cohorts redistribution (as in Caselli, Peracchi, Balbi, and Lipsi 2003) by distinguishing workers by gender and occupation.

The paper proceeds as follows. Section 2 illustrates the institutional framework, section 3 describes the projected cohort- and gender-specific mortality tables. Sections 4 and 5 describe the model and the actuarial indicators. Section 6 shows the main results, section 7 provides a sensitivity analysis with respect to some key model assumptions and section 8 concludes. Three appendixes describe (A) the methodology used to forecast mortality, (B) the formulas used to predict conversion coefficients for future retirees and (C) detailed simulation results.

2 Institutional framework

Until the 1990s, the Italian pension system was financially unsustainable. Pension expenditure grew from 7.4 % of GDP in 1970 to 14.9 % in 1992 (Brugiavini and Galasso 2004). To improve the social security budget, an impressive sequence of reforms was introduced during the 1990s. Most of them (such as the 1992 and 1997 reforms) were designed to be effective in the short run by acting on specific parameters of the existing system, such as the minimum retirement age. Law n.335/1995, which was designed instead to improve the budget in the long run, replaced the existing DB scheme with an (N)DC one. In line with its long run view, the law classified workers into three groups: the oldest, the middle-aged and the youngest. The oldest are those workers who, at the time of the reform, had accrued more than 18 years of seniority. They were left totally untouched by the reform and the old DB system is still applied to them. The middle-aged are those who had accrued less than 18 years of seniority by the end of 1995. Their pension is computed according to a mixed (*pro-rata*, PR henceforth) system where old and new rules are combined in proportion to the number of years worked prior to and after 1995. The youngest, to whom the new system fully applies, are those who started working after the 1995 reform.

In this study, we focus on private sector employees enrolled in the FPLD (*Fondo Pensioni Lavoratori Dipendenti*) fund. This fund enrols almost all employees in the private sector; in 2008, it paid around 10 millions of pensions (INPS 2009). It is managed by INPS (*Istituto Nazionale della Previdenza Sociale*), the most important social security institution in Italy. The main retirement options for FPLD workers are the old-age and the seniority pensions. Their access has been progressively tightened by various reforms during the 1990s and the 2000s. Currently, male (female) FPLD workers can claim an old-age pension at age 65 (60) provided that they have accrued 20 years of seniority. They can claim a seniority pension at age 60, provided that 35 years of seniority have been accrued and only if the sum of seniority plus age is greater than or equal to 96 (called "quota 96"). Alternatively, they can claim a seniority pension if 40 years of seniority have been accumulated. In the future, DC benefits could be claimed when 5 years of seniority have

ity plus a pension benefit equal to at least 1.2 times the social assistance benefit will be accumulated.⁴

The DB benefit is computed as the product of three factors: pensionable earnings, seniority and annual return. Pensionable earnings are the average wage of the last years of work. The number of years to include in the computation of pensionable earnings was progressively increased from 5 to 10 by the reforms of the 1990s. Seniority includes the number of years of regular contribution to the scheme, as well as years of notional contribution spent during out-of-work periods (e.g. unemployment spells, maternity leaves and military service); total seniority is topped at 40 years. Annual return is a decreasing function of pensionable earnings, equal to 2 % for a large part of the earnings distribution. Payroll tax rates grew dramatically in the last decades: from 19 % in 1967 (the first relevant year for our simulation, see section 4) to 33 %; they are paid for one third by the employee and for two thirds by the employer. Since 1992, pensions have been price-indexed.⁵

The DC pension for a worker retiring at age x is computed as:

$$P(x) = \left[c_a + \sum_{i=1}^{a-1} c_i \prod_{j=i}^{a-1} (1 + \bar{g}_j)\right] \delta_x$$
(1)

where c_i are the contributions paid when seniority is *i*, *a* is seniority at retirement and \bar{g}_j is the geometric mean of nominal GDP growth rate in the 5 years preceding the year in which seniority is *j*. The amount in squared brackets is the nominally accrued fund at retirement. δ_x is the conversion coefficient for retirement at age *x*, where $x \in [57, 65]$, defined as⁶:

$$\delta_x = \left(\frac{\sum\limits_{s=m,f} dir_{x,s} + ind_{x,s}}{2} - k\right)^{-1}$$
(2a)

$$dir_{x,s} = \sum_{t=0}^{\Omega-x} \frac{\ell_{x+t,s}}{\ell_{x,s}} (1+g_f)^{-t}$$
(2b)

$$ind_{x,s} = \theta \sum_{t=0}^{\Omega-x} \frac{\ell_{x+t,s}}{\ell_{x,s}} \left(1 - \frac{\ell_{x+t+1,s}}{\ell_{x+t,s}}\right) \left(1 + g_f\right)^{-(t+1)} a_{x+t+1}^W$$
(2c)

where $\frac{\ell_{x+t,s}}{\ell_{x,s}}$ is the gender-s-specific conditional survival probability at age x + t; g_f is the long-run expected GDP growth rate; a_{x+t+1}^W is the expected present value of a unitary annuity paid to the widow(er) at time x + t + 1; θ is the fraction of the annuity paid to the widow(er) and k is an actuarial adjustment factor that takes into account different frequencies in pension payments (set to 0.4615 by law to account for anticipated monthly payments). Notice that (equation 2a) differences in death probabilities between genders

⁴ The requirement on the minimum pension is not needed if retirement occurs at age 65.

⁵ A formal description of the DB and PR pension formulas applied to FPLD workers is provided by Fornero and Castellino (2001) and Ferraresi and Fornero (2000). Brugiavini and Galasso (2004) provide a comprehensive description of social security reforms in Italy.

⁶ Retirement at ages greater than 65 implies the application of δ_{65} and is thus highly discouraged. Retirement before age 57 is allowed only if 40 years of seniority are accrued; δ_{57} is applied in this case.

are averaged-out. Further computational details on the conversion coefficients are provided in Appendix B.

Conversion coefficients enclosed to the 1995 law incorporated the ISTAT (Italian National Institute of Statistics) 1990 mortality tables and a value for g_f equal to 1.5 %. The 1995 reform established that conversion coefficients had to be revised every ten years according to updated mortality tables and projected GDP growth rates. However, the first update of the coefficients was delayed: the new values - expected in 2005 - were legislated in 2007 (law n.247/2007) and applied in 2010. In addition to updating conversion coefficients, the 2007 law reduced their temporal validity from ten to three years. Current values, based on the ISTAT 2002 mortality tables and $g_f = 0.015$, are thus valid for the period 2010-2012.

3 Projected cohort-specific mortality tables

Starting from the 1970s, a sizable downward trend in old-age mortality has taken place in Italy.⁷ Between 1970 and 1990, male (females) life expectancy at age 60 (e_{60}) gained 2.4 (3.2) years (Caselli, Peracchi, Balbi, and Lipsi 2003) while, between 1990 and 2007, it earned an additional 3.4 (3) years (ISTAT 2010). In 2007, e_{60} was 21.89 years for males and 25.98 for females.

Positive past data on mortality has brought about an optimistic view of its future evolution. The greatest improvements are expected for the oldest-old, i.e. individuals aged 80 and over. The twenty-first century is considered by the experts as characterized not only by further progress in the prevention and cure of disease but also by firmly-rooted life styles able to promote more general "successful ageing". In this context, younger cohorts are expected to live longer than older ones, i.e. the "cohort" effect (Caselli 1990) will continue to play its positive role.

In this study, we build projected life tables for the Italian cohorts who will face the transitional phase from DB to NDC pension rules (i.e cohorts 1945-2000). We rely on the theory of mortality expansion (Myers and Manton 1984, Olshanski, Carnes, and Cassel 1993) according to which the average life span will continue to rise in the future. In order to build "limit" life tables set in the distant future, we follow the indications provided by recent interdisciplinary studies which have depicted the most important traits for human survival (see Robine, Crimmins, Horiuchi, and Zeng 2006)⁸. We then connect current tables with limit mortality tables; therefore, in the first years of the projections, our tables reflect the characteristics of current mortality, while for farther projection years, our tables progressively come closer to those of the limit scenario. The projected period is extremely long (the limit scenario is set beyond year 2100) and thus our age-timestime matrix includes all the information related to the future process of mortality of the

⁷ In the 1970s, this phenomenon was a novelty for males, since their life expectancy at age 60 had only increased by 0.25 years in the previous forty years.

⁸ Robine, Crimmins, Horiuchi, and Zeng (2006) collects studies from biology, medicine, epidemiology, demography, sociology, and mathematics. We also refer to the proceedings of the conference "Health, ageing and work. Strategies for the new welfare society in the larger Europe", The Geneva Association, Trieste, 2004.

	Ma	ales	Fem	ales
Forecast/Year:	2030	2050	2030	2050
Our tables	81.8	83.2	86.1	88.7
\mathbf{ISTAT}^a - central	82.2	84.5	87.5	89.5
\mathbf{ISTAT}^b - low	80.2	81.9	85.7	87.2

Table 1. predicted period life expectancies at birth: our tables versus official tables

a Source: ISTAT (2008)

^b Source: ISTAT (2008)

cohorts 1945-2000. Of course, for the oldest cohorts, the forecasts refer only to mortality at older ages, while for the youngest they cover their entire life. Consequently, projection uncertainty is higher for younger cohorts than for those approaching retirement in the coming years. Unfortunately, the method followed in the projections is deterministic and thus it does not allow for a quantification of projection uncertainty.⁹ Appendix A provides more methodological details.

Table 1 compares the estimated life expectancy at birth with the corresponding statistics provided by the most recent ISTAT projections (ISTAT 2008).¹⁰ The table indicates that our forecasts lie between ISTAT-low and ISTAT-central mortality scenarios. For instance, according to our tables, in 2030 male e_0 is expected to equal 81.8 years, while official statistics predict either 80.2 (low scenario) or 82.2 (central scenario). Figures 1 and 2 show the forecasted survival curves by cohort for females and males respectively. They highlight two typical characteristics of the mortality process of younger cohorts: the "rectangularization" of life tables (i.e. deaths are concentrated around a narrower interval of age) and the increase in life span. As a consequence of these improvements, younger cohorts are expected to experience sizable improvements in life expectancy: males e_{60} will increase from 23.1 years (cohort 1945) to 27.1 years (cohort 1970), up to a striking 31.4 years (cohort 2000). Corresponding values for females are 28.3, 31.6 and 38.1 years.

No official cohort life tables for the overall population exist in Italy. There exist two cohort life tables (called "RG48" and "IP55", pertaining to the 1948 and 1955 cohorts) used by Italian insurance companies to compute premiums. They refer to the annuitant population and correct for self-selection. Due to self-selection of healthier individuals, the annuitants' expected life expectancy is longer than the whole population's.

⁹ Maccheroni and Barugola (2010) project mortality for the Italian birth cohorts 1950-2005 using a Lee Carter (i.e. stochastic) model. They show that uncertainty is pretty small if one concentrates on mortality at older ages, especially for cohorts born prior to 1975.

¹⁰ At an intermediate step, our approach produces a forecast for fictitious cohorts, i.e. calculated year by year, as in the case of most forecasts. The results obtained in this phase can thus be compared with those obtained by other recent official forecasts. It should be pointed out, however, that the methodological approaches used in a very long-term forecast (like those adopted for the cohort tables we provide) and in standard demographic forecasts (which usually cover a period of 30-50 years) are extremely different. Therefore, a formal comparison between the two tables cannot be conducted.



Figure 1. Projected survival curves by cohort: females

Figure 2. Projected survival curves by cohort: males



4 Representative agents

In this exercise, we evaluate the actuarial features of the Italian pension system for a set of representative individuals subject to DB, PR or (N)DC pension rules. Each analysed individual represents a typical INPS-FPLD (i.e. private sector) employee born in a given year, of a given gender and occupation (blue and white-collar). Agents are characterized by a stylized working career, described by an age of enrolling, a lifetime wage profile and a set of alternative retirement ages. The main features of the working career (e.g. age of entry into the labour market) are kept constant across cohorts to better highlight the actuarial impact of both normative and mortality changes.

Based on the data provided in the Bank of Italy's "Survey of Household's Income and Wealth" (SHIW), our study assumes that white-collar workers enrol to the pension scheme at age 24, while blue-collar workers enrol at age 22 if female or at age 21 if males. We further assume that, once enrolled, agents keep contributing to the same scheme until retirement.¹¹ Retirement occurs between age 60 and 65. Given our assumptions, all agents become eligible to claim pension benefits at age 60 (see section 2).

Lifetime wage profiles are obtained as predictions from a random effects model for individual wages estimated on the Italian administrative data "Estratti Conto INPS", a panel data set that covers the period 1985-1997 and includes 1/365 of the Italian private sector workforce. In the wage model, the log of wages is regressed against an age spline, a variable capturing the cohort effect, a set of year, sector and area of work dummies.¹² The model is estimated separately by gender and occupation (blue and white collars). Due to self-selection problems, the estimation samples only include individuals younger than age 60. Between ages 60 and 65, we assume that wages are constant. Figure 3 displays the estimated lifetime wage profiles: among the four groups, male white-collar workers have the steepest and highest profile while female blue-collar workers are the poorest and have the less dynamic careers.

Given these stylized careers, each representative agent is assigned to a specific pension regime (DB, PR or NDC). Table 2, for example, reports assigned pension regime and retirement years by cohort for white-collar employees. It shows that agents born between 1945 and 1953 are subject to the DB rules, since in 1995 they had accumulated 18 years of contributions or more. Depending on retirement age and cohort, they claim pension benefits between year 2005 and 2018. Employees born between 1954 and 1970 are subject to the PR rules and retire between 2014 (cohort 1954, retiring at age 60) and 2035 (cohort 1970, retiring at age 65). Finally, employees born in 1971 or later are subject to the DC rules and retire in 2031 or later. A similar categorization of workers (not shown in the table) is obtained for male blue-collar and female blue-collar workers.

For each agent and retirement age between 60 and 65, we compute the accrued pension benefit and corresponding actuarial indicators (see section 5). In these computations, we consider both historical and projected values of key macroeconomic indicators such as GDP growth rates, inflation rates and long-run (10 years' maturity) government bonds interest rate. Historical values are taken from various official sources: the OECD, the Bank

- ¹¹ Using the SHIW data (various cross-sections) we compute gender- and occupation-specific average ages of entry into the labour market. To avoid overestimation of accrued seniority at retirement, we account for an average period of 3 years spent out of the labour market and not compensated by notional contributions (see section 2).
- ¹² Given the long-run horizon of our analysis, we also need to predict wages for out-of-sample cohorts. Traditional econometric models, which use dummies to capture cohort effects, require additional assumptions for out-of-sample predictions. Following Heckman and Robb (1985) and Kapteyn, Alessie, and Lusardi (2005) we assume that wages differ across cohorts due to the macroeconomic conditions when individuals enter into the labour market. These conditions are summarized by productivity growth and are approximated by GDP per capita. Econometric details are provided upon request.



Figure 3. Estimated age-wage profiles by gender and occupation

Note: log-wages in $\in 10,000$ (2009 euros).

of Italy and ISTAT. In our main macroeconomic scenario, we assume a projected long-run real GDP growth rate (g_f , see section 2) equal to 1.5 % and a long-run riskless interest rate (r, see section 5) equal to 2 %. A very similar value for the GDP growth rate has been adopted in recent long-run EPC-WGA pension projections (EPC-WGA 2008) and in most of the Italian pension expenditure projections (MEF 2009, MEF 2011).¹³ The implied spread ($r - g_f = 0.5$ percentage points) is coherent with a dynamic efficient economy although it does not overestimate the dominance of the financial market with respect to the pay-as-you-go system. In a sensitivity analysis we consider alternative macroeconomic scenarios where the spread between the two key macroeconomic variables is wider.¹⁴

Actuarial fairness is evaluated in three different normative scenarios: "baseline 1995 (B95)", "baseline 2007 (B07)" and "quasi-actuarial benchmark (AB)". The first two incorporate actual pension rules. Scenario B95 considers the 1995 pension rules, whereas scenario B07 takes into account the reduced validity (from 10 to 3 years, as modified by the 2007 law) of the conversion coefficients. The AB scenario considers a hypothetical quasi-actuarial fair pension system, where conversion coefficients are cohort- and gender-specific and computed according to equation (B 2) in appendix B.¹⁵

¹³ See also MLSP (2002a) and MLSP (2002b). As already mentioned in section 2, a value of 1.5 % for the GDP growth rate has been used to compute conversion coefficients in 1995 and 2007.

¹⁴ Recent long-run EPC-WGA pension projections (EPC-WGA 2008) consider an interest rate equal to 3 %, which implies a wider spread $r - g_f$.

¹⁵ Consistently with current rules, in the B95 and B07 scenarios, conversion coefficients for the DC and PR benefits incorporate cross-sectional mortality rates, which we obtain by looking at the appropriate column of the gender-specific age-times-year projected life table (see equation (B1))

 cohort	\mathbf{start}^a	$sen.1995^b$	\mathbf{regime}^{c}	retirem age 60	nent years: age 65
1945	1969	26	DB	2005	2010
1953	1977	18	DB	2013	2018
1954	1978	17	\mathbf{PR}	2014	2019
 1970	 1994	 1	\mathbf{PR}	 2030	2035
 1971	1995	0	DC	2031	2036
 2000	 2024	 0	DC	 2060	2065

Table 2. pension regime and retirement years by cohort: white-collar workers

^a enrollment year.

 b seniority accrued at the end of 1995.

 c DB=defined benefit, PR=pro-rata, DC=notional defined contribution.

5 Actuarial indicators

We evaluate the actuarial characteristics of the Italian pension system by means of two social security money's worth measures (Geanakoplos, Mitchell, and Zeldes 2000): the present value ratio (PVR) and the implicit tax/subsidy rate (TAX). The former is used to evaluate actuarial fairness while the latter measures actuarial fairness at margin. Both of them require the computation of social security wealth (SSW). The SSW for retirement at a, computed at $t_1 \in [1, a + 1]$ and evaluated at t_2 , where a, t_1, t_2 define years of seniority, is given by:

$$SSW^{a}_{t_{1},t_{2}} = -\sum_{j=t_{1}}^{a} c^{*}_{j} (1+r)^{t_{2}-j} + \left[P(e+a) \frac{1}{\overline{\delta}^{co}_{e+a,s}} \right] (1+r)^{t_{2}-(a+1)}$$
(3)

where c_j^* are the contributions at constant prices paid by the worker to the fund when accrued seniority is j and r is the time-constant riskless interest rate.¹⁶ P(e + a) is the pension benefit associated with retirement at age e + a, where e is the age at which the employee starts contributing to the scheme and a is the number of years of seniority accrued at retirement. $\overline{\delta}_{e+a,s}^{co}$ is equal to $\delta_{e+a,s}^{co}$ as described in equation (B 2) but with g_f replaced by r. The sum in square brackets is therefore the present value of expected

and section 3). In the AB scenario, mortality rates are selected from the appropriate diagonal of the gender-specific age-times-year projected life table.

¹⁶ In the empirical analysis, we approximate r with the long-run (10 years' maturity) government bonds interest rate. See Queisser and Whitehouse (2006) for a discussion of the appropriate discount rate. Notice that (see e.g. Coile and Gruber 2000) we assume that the individual is alive at retirement. pension benefits. It is computed assuming cohort- and gender- specific mortality rates, a discount rate equal to r, price-indexation of pensions and survivors' benefits. Financial flows are annual and anticipated.

The PVR is given by the ratio between the present value of expected pension benefits and the present value of the contributions paid during the working career. This indicator shows how much the system returns to the worker for each euro paid. The PVR computed when seniority at retirement is equal to a is given by:

$$PVR^{a} = \frac{P(e+a)\frac{1}{\overline{\delta}_{e+a,s}^{co}}}{\sum_{j=1}^{a} c_{j}^{*}(1+r)^{a+1-j}}$$
(4)

A pension system is defined as actuarially fair if $PVR^a = 1$ (i.e. $SSW_{1,1}^a = 0$ in equation 3). A pension system is defined as *quasi*-actuarially fair if $PVR^a = 1$ when $r = g_f$.

The TAX computed at $t_1 \in [1, a + 1]$, once a' years of seniority have been accrued and evaluated at t_2 , is given by:

$$TAX_{t_1,t_2}^{a'} = \frac{-acc_{t_2}^{a'}}{E_{t_1} \left[w_{a'+1} \right] (1+r)^{t_2 - (a'+1)}}$$
(5)

where the numerator is called *accrual* and is defined as:

$$acc_{t_{2}}^{a'} = SSW_{t_{1},t_{2}}^{a'+1} - SSW_{t_{1},t_{2}}^{a'}$$

$$= -c_{a'+1}^{*}(1+r)^{t_{2}-(a'+1)} + \left[P(e+a'+1)\frac{1}{\bar{\delta}_{e+a'+1,s}^{co}} - P(e+a')\frac{1}{\bar{\delta}_{e+a',s}^{co}}(1+r) \right]$$

$$(1+r)^{t_{2}-(a'+2)}$$

$$(6)$$

Equation (6) highlights that, if retirement is postponed by one year, the SSW varies due to two reasons. First, it decreases because of additional contributions to pay. Second, it varies due to the difference in square brackets between the present values of pension benefits associated with the alternative retirement options a' and a' + 1. The sign of this difference is undefined because a shorter retirement period is generally associated with a higher pension benefit. In equation (5), the accrual is normalized with respect to the expected wage for the additional year of work.

A pension system is defined as actuarially fair at margin if $TAX_{t_1,t_2}^{a'} = 0$. A pension system is defined as *quasi*-actuarially fair at margin if $TAX_{t_1,t_2}^{a'} = 0$ when $r = g_f$ in (5). If $TAX_{t_1,t_2}^{a'} > 0$ (i.e. $Accr_{t_2}^{a'} < 0$) the pension system imposes an implicit taxation on the continuation of the working activity, thus providing financial incentives to early retirement; if $TAX_{t_1,t_2}^{a'} < 0$ the pension system penalizes early retirement.

6 Results

6.1 Forecasted conversion coefficients

Table 3 shows legislated and forecasted conversion coefficients by retirement age (57-65) and selected retirement years. In the first two columns, it reports values fixed by the 1995

Table 3. Legislated and forecasted conversion coefficients by age and retirement year: current legislation

		Selected r	etirement	years (life	tables)	
	1995-09	2010-12	2019-21	2028 - 30	2040-42	2049-51
Age	(ISTAT90)	(ISTAT02)	(2018)	(2027)	(2039)	(2048)
57	4.720	4.419	4.169	4.029	3.866	3.793
58	4.860	4.538	4.276	4.128	3.956	3.879
59	5.006	4.664	4.390	4.234	4.052	3.971
60	5.163	4.798	4.511	4.345	4.153	4.067
61	5.334	4.940	4.640	4.464	4.261	4.170
62	5.514	5.093	4.777	4.590	4.375	4.279
63	5.706	5.257	4.922	4.724	4.496	4.394
64	5.911	5.432	5.077	4.867	4.625	4.517
65	6.136	5.620	5.244	5.020	4.762	4.648
	pe	ercentage dev	viation wit	h respect	to 1995-09)
57	-	-6.38	-11.67	-14.64	-18.10	-20.38
58	-	-6.63	-12.01	-15.06	-18.60	-20.93
59	-	-6.83	-12.30	-15.43	-19.06	-21.45
60	-	-7.07	-12.62	-15.83	-19.56	-22.01
61	-	-7.39	-13.01	-16.31	-20.12	-22.63
62	-	-7.64	-13.37	-16.75	-20.66	-23.23
63	-	-7.87	-13.74	-17.20	-21.21	-23.84
64	-	-8.10	-14.10	-17.66	-21.76	-24.45
65	-	-8.41	-14.54	-18.19	-22.39	-25.14

Note: percentage points; the first two columns report coefficients determined by laws n.335/95 and n.247/07; the following columns report forecasted coefficients (scenario B07) for selected triennial retirement periods. Coefficients are computed using forecasted cross-sectional life tables for the year indicated in italics at the top of each column.

and 2007 laws and valid respectively for the period 1995-2009 and 2010-12. In the following columns, the table shows forecasted coefficients (scenario B07) for selected retirement periods (2019-21, 2028-30, 2040-42, 2049-51). Conversion coefficients are expected to decrease considerably, as a consequence of the increased longevity. For example, the coefficients which will be applied in 2019-21 are expected to be 12-15 % lower than those legislated in 1995 (see lower part of the table). These findings are in line with those found by Caselli, Peracchi, Balbi, and Lipsi (2003) in their low-mortality scenario.¹⁷ The bottom part of table 3 shows that the relative reduction in the coefficients is higher for older retirement ages and concentrated in the first years of the simulation.

Table 4 reports the forecasted conversion coefficients according to the AB scenario for

¹⁷ Caselli, Peracchi, Balbi, and Lipsi (2003) predict conversion coefficients for 2020 using projected ISTAT life tables (base 2002, central and low-mortality scenarios, unpublished forecasts).

		S	selected	cohort	s	
Age	1950	1960	1970	1980	1990	2000
			Ma	ales		
57	4.044	3.896	3.769	3.656	3.557	3.466
58	4.140	3.985	3.851	3.734	3.629	3.534
59	4.240	4.079	3.939	3.816	3.706	3.606
60	4.347	4.178	4.031	3.902	3.787	3.682
61	4.459	4.282	4.128	3.993	3.872	3.762
62	4.578	4.392	4.231	4.089	3.961	3.846
63	4.704	4.509	4.339	4.190	4.056	3.934
64	4.837	4.632	4.454	4.297	4.156	4.028
65	4.978	4.763	4.575	4.410	4.262	4.127
			Fem	ales		
57	3.925	3.767	3.627	3.498	3.375	3.255
58	4.014	3.848	3.701	3.566	3.437	3.312
59	4.108	3.933	3.779	3.637	3.503	3.371
60	4.208	4.024	3.862	3.713	3.571	3.433
61	4.313	4.119	3.949	3.792	3.643	3.499
62	4.425	4.220	4.041	3.875	3.719	3.568
63	4.543	4.327	4.137	3.964	3.800	3.640
64	4.668	4.440	4.240	4.057	3.884	3.716
65	4.802	4.560	4.348	4.155	3.973	3.797

Table 4. Forecasted conversion coefficients by age, cohort and gender: AB scenario

selected cohorts. It highlights how heterogeneous the conversion coefficients would be, were they computed consistently with the cohort and gender differentials in life expectancy. For example, the coefficient applied to a male born in 1960 and retiring at age 60 ($\delta_{60,m}^{1960} =$ 4.178) would be 3.9 % lower than that applied to a male retiring at the same age but born 10 years earlier ($\delta_{60,m}^{1950} = 4.347$). The coefficient applied to a male born in 2000 would be 15.3 % lower than that applied to an employee born in 1950 (cf. $\delta_{60,m}^{2000}$ with $\delta_{60,m}^{1950}$). Lower conversion coefficients would be applied to females, since they live longer than males; the provision of a joint annuity (see equation B 2, the indirect component) would only partially compensate the difference in life expectancy between genders in the main (direct) pension beneficiary.

Finally, table 5 shows - for each gender, retirement age and selected retirement year - the percentage deviation between coefficients computed according to the current legislation (scenario B07, table 3) and those computed according to the actuarial benchmark (AB scenario, table 4). For instance, the coefficient applied by law to a male employee retiring in 2010 at age 60 (4.798, see table 3) is 10.4 % higher than the coefficient computed according to the cohort- and gender-specific mortality rates ($\delta_{60,m}^{1950} = 4.347$, see table 4). The figures in the table are positive, meaning that the rules adopted by the 1995 reform

	Se	elected	retirem	ent yea	\mathbf{rs}
Age	2010	2020	2030	2040	2050
			Males		
57	10.5	8.1	7.9	6.6	6.5
58	10.5	8.1	7.9	6.6	6.5
59	10.4	8.0	7.8	6.5	6.7
60	10.4	8.0	7.8	6.4	6.3
61	10.3	7.9	7.8	6.4	6.3
62	10.3	7.9	7.7	6.3	6.2
63	10.2	7.8	7.7	6.2	6.1
64	10.2	7.8	7.6	6.1	6.1
65	10.2	7.7	7.6	6.1	6.0
			Females		
57	14.1	12.0	12.3	11.7	12.5
58	14.1	12.0	12.4	11.8	12.6
59	14.1	12.1	12.5	11.8	12.7
60	14.0	12.1	12.5	11.9	12.8
61	14.1	12.1	12.6	11.9	12.8
62	14.1	12.2	12.6	12.0	12.9
63	14.1	12.2	12.7	12.0	12.9
64	14.1	12.2	12.7	12.0	13.0
65	14.1	12.2	12.8	12.0	13.0

Table 5. Forecasted conversion coefficients: percentage deviation between current legislation (Table 3) and AB scenario (Table 4)

=

to compute conversion coefficients - which rely on cross-sectional life tables and disregard cohort improvements in longevity - provide pension benefits which are more generous than what the benchmark suggests. This "premium" is lower for males than for females, since it is partly compensated (see equation 2a) by the application of unisex conversion coefficients averaging-out differences in mortality rates between genders.

6.2 Actuarial fairness

The PVR by cohort for a typical male white-collar worker retiring at age 60 is reported in figure 4. As specified in section 4, we consider three alternative scenarios (B07, B95 and AB), we assume a long-run GDP growth rate equal to 1.5 % and an risk-less interest rate equal to 2 %. Results for other agents, retirement ages and macroeconomic assumptions are reported in section 7.

Figure 4 highlights that the pre-reform system is extremely generous, providing much more than actuarially fair benefits. Cohorts belonging to the DB scheme (1945-1953), in fact, receive in terms of social security benefits up to 150 % of what they paid during

their working career. The PVR sharply falls (-14 percentage points, from 141 to 127 %) between the 1953 and 1954 cohorts, as a consequence of the classification of workers according to either the DB (cohort 1953) or the PR (cohort 1954) rules (see Ferraresi and Fornero 2000, Fornero and Castellino 2001). Throughout the transitional phase toward the DC rules (cohorts 1954-1970) the PVR exhibits a decreasing trend due to the increasing weight of the DC quota in the computation of pension benefits of younger cohorts. For cohorts younger than 1967, the system is less than actuarially fair. For cohorts born in 1971 and later, the indicator ranges between 89.7 and 91 %, indicating that the steady state DC regime is less than actuarially fair.

Results for the steady state can be attributed to dynamic efficiency (see Fornero and Castellino 2001): in an economy where the risk-less rate of return from financial markets dominates its rate of growth, only funded DC schemes can guarantee "pure" actuarial fairness. Regardless of macroeconomic conditions, a comparison across scenarios shows that the Italian pension system does not meet the weaker requirement of *quasi*-actuarial fairness on average (Palmer 2006). The PVR in the B07 and B95 scenarios is higher (by 5-6.5 percentage points) than in the AB scenario, meaning that the steady state DC rules are more than *quasi*-actuarially fair. This result depends on the rules adopted for the computation of conversion coefficients: since cohort effects in mortality are disregarded, retired cohorts will likely live longer than what accounted for in the computation of their pension benefit.

Finally, figure 4 shows that the PVR is characterized by an irregular course in the B07 and B95 scenarios. Periodical adjustments of the conversion coefficients, implemented to *ex-post* counteract increased longevity, generate discontinuities in the pension treatment of adjacent cohorts. The 2007 reform, by reducing from 10 to 3 years the temporal validity of the conversion coefficients, was quite effective in reducing redistribution across adjacent cohorts (cf. discontinuities in the B07 and B95 scenarios).¹⁸ The PVR has a smooth course in the AB scenario, where cohort- and gender-specific longevity increases are fully offset by corresponding reductions in conversion coefficients. For this reason, this simulation is similar to a kind of "static mortality case". Accordingly, our results for the AB scenario are very similar to those reported in previous studies (Ferraresi and Fornero 2000) which assume static mortality.

¹⁸ Consider, for example, cohorts 1971 to 1974. A male white-collar worker born in 1971 is aged 60 in 2031. Under the current rules (B07), his PVR is equal to 91.4 %. A worker born in 1972 (1973) is expected to live slightly longer than one born in 1971, while his pension benefits will be computed using the same conversion coefficient of the previous cohort. As a consequence, he will enjoy a slightly higher PVR=91.6 (91.9). A worker born in 1974 and retiring at age 60 will be hit by the revision of the coefficients fixed by law in 2034 (NPVR=91.1). However, the reduction in the PVR suffered by the 1974 cohort with respect to the 1973 cohort is small, i.e -0.8 percentage points. According to the 1995 legislation (coefficients revised in 2040), the effect of longevity improvements for 9 years combined with less frequent revisions of the coefficients would generate a much more pronounced cycle in the indicator: the PVR would increase by 3.2 points between the 1979 and 1980 cohorts due to the coefficients revision.

Figure 4. PVR by cohort at age 60: male white-collar workers, alternative scenarios



Note: percentage points; B07: current legislation; B95: conversion coefficients revised every ten years; AB: quasi-actuarial benchmark; 36 years of seniority; $g_f = 0.015, r = 0.02$

6.3 Actuarial fairness at margin

Figure 5 shows the TAX by cohort for male white-collar workers retiring at age 60. In the DB scheme the indicator reaches a value of 40 %, which represents a strong financial incentive to claim pension benefits at minimum requirements (see e.g. Belloni and Alessie 2009). The application of PR rules and the full application of the DC formula drastically reduce the implicit taxation of continuing working. However, the taxation is not eliminated in the steady state, which is less than actuarially fair at margin for most of the cohorts: excluding retirement years in which conversion coefficients are updated (see next paragraph), the indicator ranges between 3.8 and 4.1 %. Due to dynamic efficiency, even in the DC scheme, it is optimal to retire at minimum requirements and invest the accumulated wealth at the market return $r > g_f$.

Figure 5 (B07) points out the existence of big spikes of TAX (equal to about 11-12 percentage points) for the years in which conversion coefficients are updated. An individual considering staying at work in the year before the revision of the conversion coefficients would face a very strong constraint: his social security wealth would be considerably reduced if he kept working.¹⁹ The 2007 reform, by restricting the time validity of the

¹⁹ Compare, for example, workers born in 1972 and 1973. The former are aged 60 in 2032. They face a TAX equal to 3.8 % ($\delta_{60} = 4.295$ in 2032, they expect $\delta_{61} = 4.41$ in 2033). Workers born in 1973 face the revision of the coefficients when they reach age 60 and are taxed at a rate equal to 16.8 % if they postpone retirement by one year (they expect δ_{61} to fall from 4.41 to 4.36 due to the revision in 2034).

Figure 5. TAX by cohort at age 60: males white-collar workers, alternative scenarios



Note: percentage points; B07: current legislation; B95: conversion coefficients revised every ten years; AB: quasi-actuarial benchmark; 36 years of seniority; $g_f = 0.015, r = 0.02$

conversion coefficients to three years, reduced the implicit taxation in a sizable way (the B95 scenario shows spikes equal to 38-45 percentage points, cf. spikes in the B07 scenario).

A comparison across scenarios shows that the Italian NDC system is more than quasiactuarially fair at margin, meaning that it generates an implicit tax which is lower than that determined by the spread $r - g_f$. Indeed, in the B07 scenario, the TAX is 4-5 percentage points lower than in the AB scenario (excluding the spikes). This "extra-return" embedded in the current rules is due to the upward-biased difference between the conversion coefficients applied to adjacent retirement ages, which reflects the application of cross-sectional mortality rates in their construction. In the AB scenario, the TAX has a smooth trend since, in this simulation, longevity differences by cohort are compensated by appropriate differences in conversion coefficients applied to different cohorts.

7 Sensitivity

In this section, we provide a sensitivity analysis of the main results illustrated in the previous section comparing findings for alternative retirement ages (60-65), different agents (male blue-collar workers and females) and different macroeconomic settings. The full set of results for the base macroeconomic case is reported in Appendix C.

In the DB scheme, results show a negative relationship between the PVR and the retirement age whereas in the DC scheme the indicator is almost invariant with respect to retirement age. To better understand these findings, in Figure 6 we report the PVR as well as its components - present value of benefits (PVB) and present value of contributions

Figure 6. PVR, PVB and PVC for alternative retirement ages, cohorts 1945 (DB) and 2000 (DC)



Note: PVR left axis; pvb and pvc=1,000 euros, prices 2009, right axis; male white-collar workers; current legislation (B07); $g_f = 0.015, r = 0.02$

(PVC), see equation 4 - for two selected cohorts (1945, DB and 2000, DC) retiring at ages 60-65. For the DB cohort, the negative impact on the PVB of the shorter retirement period associated with later retirement dominates over the positive effect related to the modest increase in the pension benefit (2 % for each additional working year up to 40 years of seniority, plus the increase in pensionable earnings). Therefore, both the PVB and the PVR are lower for later retirement ages. For the DC cohort, the (actuarially-related) increase in the pension benefit associated with later retirement is more pronounced than in the DB case, and the consequent increase in the PVB almost perfectly compensates the increase in the PVC.²⁰ The TAX rate (see tables in Appendix C), on the contrary, tends to increase with the age of retirement in the DB scheme, thus providing increasingly stronger incentives to retire. The indicator becomes especially high when the individual accrues 40 years of seniority. In this case, in fact, there is a no annual return for the additional years of work. The TAX is instead almost constant with respect to the retirement age in the DC scheme.

²⁰ A lower PVR associated with older retirement ages does not allow us to infer that there is an incentive to retire (and viceversa). For this kind of considerations, we have to refer to the TAX. As already explained in section 5 (see equation 5) both the SSW associated with immediate retirement and the expected SSW associated with retirement in one year are evaluated at the same point in time in the TAX. When we compare PVR associated with different retirement ages, we instead refer to computations made at different points in time and based on currently accrued pension benefits.

As regards the comparison across genders and occupations, it turns out that the 1995 reform changed the implicit redistribution of the Italian pension system from "wagebased" to "mortality-based". In the DB scheme, the actuarial indicators are particularly affected by the shape of the lifetime wage profile. Figure 7 reports the PVR by cohort for retirement at age 60, for four types of agents: male blue-collar, male white-collar, female blue-collar and female white-collar workers. In the DB scheme, the flattest agewage profiles are typically associated with lower PVR. If, for instance, we consider the 1945 cohort, the indicator is equal to 154 % for male white-collar workers (steepest profile) and to 126.5 % for female blue-collar workers (flattest profile). The flattest lifetime wage profiles also typically correspond to a higher implicit taxation: the corresponding TAX for the above reported examples are equal to 37.7 (male white-collar) and 64.3 % (female blue-collar). These results are explained by the DB pension formula, which accounts for the wages in the last part of the working career only. By postponing retirement, the implicit taxation is then particularly high for a flat age-wage profile, since in this case pensionable earnings do not increase. In the (N)DC scheme, the indicators are affected by group-specific (in our exercise, gender-specific) mortality rates. Figure 7 illustrates that the PVR is approximately 90 % for males and around 95 % for females. Females are favoured, since their longer life expectancy is not accounted for in the computation of conversion coefficients.²¹

Finally, we report the results of a sensitivity analysis with respect to the macroeconomic scenario. Note that the sign of $\Delta PVR/\Delta i$ and $\Delta TAX/\Delta i$, $i = \{r, g_f\}$, in both the DB and the DC scheme is known *a priori* (see section 5) and is reported in the upper part of table 6 ("Qualitative impact"). While *r* affects the results in both the DB and the DC scheme, g_f affects them only in the DC scheme. To provide a quantification of these effects, we simulate the PVR and the TAX under a number of different values for g_f and *r*. We restrict our analysis to the sets $r \in [0.01, 0.03]$ and $g_f \in [0.005, 0.025]$, s.t. $r > g_f$. We consider two "extreme" cohorts: 1945 (DB) and 2000 (DC).

Simulated results for male white-collar workers retiring at age 60 are illustrated in detail in figure 8 and summarized in the bottom part of table 6. Each panel in figure 8 shows the value assumed by a given indicator and for a given cohort, as a function of the two macro variables. Each (iso-)line displays the set $\{g_f, r\}$ such that the indicator has the same value; each line refers to a decile of the simulated PVR or TAX distribution: lighter lines are associated with a higher value of the indicator. In the bottom part of table 6 ("Simulation results"), we report the corresponding 25th, 50th and 75th percentiles of the simulated PVR and TAX distributions.

Results in figure 8 are fully consistent with the expected effects described above. We refer to the upper part of table 6 for their interpretation. The simulated PVR (TAX) tends

²¹ In the DC scheme, the TAX is slightly negative for females (see appendix C), while it is slightly positive for males. As for males, two effects counteract: dynamic efficiency, which tends to increase the tax, and the upward-biased difference in the conversion coefficients applied to adjacent retirement ages, which leads to a reduction in the TAX (the "extra return" described in 6.3). As compared to males, females benefit from favourable gender-neutral conversion coefficients. Therefore, the latter effect is stronger for females, which explains the difference in the sign of the TAX between the two genders.



Figure 7. PVR by cohort at age 60: current legislation, different agents

Note: scenario B07; $g_f=0.015, r=0.02$

to be lower (higher) than in the base case, since in most cases, the spread $r - g_f$ is greater than 0.5 percentage points.²² Empirical evidence shows that the interquartile range turns out to be quite restricted, especially for the TAX (e.g. 43.9-35.5=8.4 percentage points for the 1945 cohort). The reference thresholds for the two indicators, i.e. 1 for the PVR and 0 for the TAX, are almost never "crossed" in either the DB or the DC scheme. Overall, the main conclusions derived for the base case do not seem to change dramatically for a wide range of macroeconomic conditions.

²² In the base case ($g_f = 0.015$ and r = 0.02), PVR is equal to 154.1 for the 1945 cohort and to 90.2 for the 2000 cohort, while TAX is equal to 37.5 and 14.1 respectively (see Appendix C).

	1945	Cohort ((DB)	(scheme) 2000	(DC)
		Qualitatia	ve impact:	
	$\Delta \ \mathrm{PVR}$	Δ TAX	$\Delta \mathrm{PVR}$	Δ TAX
Δr	-	+	-	+
Δg_f	0	0	+	-
		Simulatic	on results:	
percentiles:	PVR	TAX	PVR	TAX
25	126.2	35.5	69.7	13.3
50	141.4	40.3	80.4	17.3
75	163.2	43.9	92.3	20.9

Table 6. Sensitivity analysis with respect to g_f and r: qualitative impact and simulation results

Note: percentage points; male white-collar workers retiring at age 60 in the B07 scenario; $r \in [0.01, 0.03], g_f \in [0.005, 0.025], r > g_f, 21 \times 21$ simulations

Figure 8. Sensitivity analysis with respect to g_f and r: isolines of PVR and TAX for the 1945 and 2000 cohorts



Note: male white-collar workers retiring at age 60 in the B07 scenario; $r \in [0.01, 0.03], g_f \in [0.005, 0.025], r > g_f, 21 \times 21$ values; each (iso-)line display the set $\{g_f, r\}$ such that the indicator gets the same value; each line refers to a decile of the simulated PVR or TAX distribution; lighter lines are associated with an higher value of the indicator.

8 Conclusions

In this study, we analysed the actuarial features of the Italian pension system after the 1995 reform. We considered both the pre-reform DB, the *pro-rata* and the new (N)DC rules applied to INPS-FPLD (i.e. private sector) employees. Although our analysis focuses on actuarial differences across cohorts, we also provide some evidence of within cohorts redistribution by distinguishing workers by gender and occupation. We computed two actuarial indicators, namely the present value ratio (PVR) and the implicit tax rate (TAX). Improving on the existing literature (Ferraresi and Fornero 2000, Fornero and Castellino 2001), we allowed for dynamic mortality in their computation. We particularly investigated the actuarial properties of the rules designed by the 1995 reform (as modified by the 2007 law) to handle increasing longevity. To this aim, we built projected cohort- and gender-specific mortality tables based on a limit demographic scenario recently depicted by demographic experts. Results for the current legislation were compared with those stemming from a scenario where conversion coefficients, used to transform the accumulated sum at retirement into the annuity, are based on cohort- and gender-specific mortality rates.

The old DB pension rules provide extremely generous, much more than actuarially fair, pension benefits. The steepest age-wage profiles are typically associated with higher PVR. If, for instance, we consider the 1945 cohort, the indicator is equal to 154 % for male white-collar workers (steepest profile) and to 126.5 % for female blue-collar workers (flattest profile). The DB scheme also provides strong incentives to early retirement (TAX up to 70 % for female blue-collar workers). Throughout the transitional phase toward the DC rules, the generosity and the implicit taxation of the system progressively decrease.

Due to dynamic efficiency, the steady-state (N)DC regime is less than actuarially fair (the PVR is about 90 % for males and 95 % for females). For the same reason, the new regime is less than actuarially fair at margin for most of the retiring cohorts and agents (excluding the years in which conversion coefficients are updated, the TAX is about 4 %). The implicit redistribution of the Italian pension system changed from "wage-based" (DB) to "mortality-based" (NDC).

As a consequence of the rules adopted to compute conversion coefficients and to update them in response to increasing longevity, the future DC steady state is more than *quasi*actuarially fair on average (at margin). Periodical *ex-post* adjustments in the conversion coefficients generate discontinuities in the pension treatment of adjacent cohorts as well as strong incentives to retire prior to each adjustment. The 2007 reform, reducing from 10 to 3 years the temporal validity of the conversion coefficients, was quite effective in improving actuarial fairness. Nevertheless, actuarial unfairness is still embedded in the Italian DC system, a problem that mostly depends on the use of historical cross-sectional mortality rates in the computation of conversion coefficients. Retired cohorts will likely live longer than what accounted for in the computation of their pension benefits. This discrepancy occurs because cohort effects in mortality are disregarded by the Italian law.

Finally, our simulations highlight that adopting cohort-specific conversion coefficients based on projected mortality tables would drastically improve actuarial fairness across cohorts and retirement ages. Since in that case pension computations would be based on mortality projections, the implementation of such a policy would mean dealing with both possible discretionary policy interventions and uncertainty. For this reason, operations and proceedings by the demographic experts in charge of producing mortality tables would need to be as transparent as possible. Revisions would need to be performed by a strictly technical team. Uncertainty related to the projections can be indeed high for young cohorts, but is more limited for those approaching retirement (unfortunately, we cannot quantify projection uncertainty in our exercise). As Palmer (2006) pointed out, quasi-actuarially fair NDC pension systems can opt for either uncertainty at retirement i.e. relying on projected cohort mortality when annuitizing, but keeping constant benefits for retirees - or uncertainty after retirement - i.e. regularly adjusting the pensions of retirees on the basis of updated historical statistics. Current Italian social security rules are not a feasible quasi-actuarially fair alternative.

References

AARON, H. (1966): "The Social Insurance Paradox," Canadian Journal of Economics, 32.

- BELLONI, M., AND R. ALESSIE (2009): "The importance of financial incentives on retirement choices: New evidence for Italy," *Labour Economics*, 16(5), 578–588.
- BORELLA, M., AND F. CODA (2006): "Distributive Properties of Pension Systems: A Simulation of the Italian Transition from Defined Benefit to Defined Contribution," *Giornale degli Economisti e Annali di Economia*, 65(1), 95–125.
- BRASS, W. (1971): On the Scale of Mortality, Biological Aspects of Demography. Taylor and Francis, London.
- BRUGIAVINI, A., AND V. GALASSO (2004): "The Social Security Reform Process in Italy: Where Do We Stand?," *Journal of Pension Economics and Finance*, 3(2), 165–195.
- BRUGIAVINI, A., AND F. PERACCHI (2003): "Social Security Wealth and Retirement Decisions in Italy," *Labour*, 17, 79–114.
- (2004): "Micro-Modeling of Retirement Behavior in Italy," in Social Security Programs and Retirement Around the World: Micro-Estimation, ed. by J. Gruber, and D. Wise, NBER, chap. 6, pp. 345–398. The University of Chicago Press, Chicago and London.
- CASELLI, G. (1990): "The Influence of Cohort Effects on Differentials and Trends in Mortality," in *Measurement and Analysis of Mortality. New Approaches*, ed. by D. Vallin, and Palloni. Clarendon Press, Oxford.
- CASELLI, G., F. PERACCHI, E. BALBI, AND R. LIPSI (2003): "Differential Mortality and the Design of the Italian System of Public Pensions," *Labour*, 17(0), 45–78, Special issue.
- COILE, C., AND J. GRUBER (2000): "Social Security Incentives for Retirement," NBER Working Paper n. 7651.
- DG-ECFIN (2006): "The impact of ageing on public expenditure: projections for the EU-25 Member States on pensions, healthcare, long-term care, education and unemployment transfers," Special Report 1, European Commission and Directorate-General for Economic and Financial Affairs, Brussels.

- DISNEY, R. (2004): "Are Contributions to Public Pension Programmes a Tax on Employment?," *Economic Policy*, 19(39), 267–300.
- DUCHÊNE, J., AND G. WUNSCH (1993): "Population Aging and the Limits to Human Life," in *Future Demographic Trend in Europe and North America*, ed. by W. Lutz. Academic Press, London.
- EPC-WGA (2008): "The 2009 Ageing Report: Underlying assumptions and Projections Methodologies for the EU-27 Member states (2008-2060)," European Economy n.7, European Policy Commettee - European Commission.
- FERRARESI, P., AND E. FORNERO (2000): "Social Security Transition in Italy: Costs, Distortions and (Some) Possible Corrections," CeRP Working Paper n. 2.
- FORNERO, E., AND O. CASTELLINO (2001): La Riforma del Sistema Previdenziale Italiano. Il Mulino, Bologna.
- GEANAKOPLOS, J., O. MITCHELL, AND S. ZELDES (2000): "Social Security Money's Worth," in *Prospects for Social Security Reform*, ed. by R. M. O. Mitchell, and H. Young, chap. 5, pp. 79–151. Pension Research Council, The Wharton School, University of Pennsylvania Press, Philadelphia.
- GRUBER, J., AND D. WISE (1999): Social Security and Retirement Around the World, NBER. The University of Chicago Press, Chicago and London.
- HECKMAN, J., AND R. ROBB (1985): "Using Longitudinal Data to Estimate Age, Period and Cohort Effects in Earnings Equations," in *Cohort Analysis in Social Research: Beyond the Identification Problem*, ed. by Mason, and Fienberg. Sprinter-Verlag, New York.
- INPS (2009): "Osservatorio sulle pensioni www.inps.it," .
- ISTAT (2008): "Previsioni Demografiche 1.1.2007-1.1.2051," http://demo.istat.it, a cura di M. Marsili and M.P. Sorvillo.
 - (2010): "Tavole di Mortalit della popolazione italiana per provincia e regione di residenza demo.istat.it," .
- KAPTEYN, A., R. ALESSIE, AND A. LUSARDI (2005): "Explaining the Wealth Holdings of Different Cohorts: Productivity Growth and Social Security," *European Economic Review*, 49, 1361–1391.
- KITAGAWA, E., AND P. HAUSER (1973): Differential mortality in the United States: a Study in Socioeconomic Epidemiology. Harvard University Press, Cambridge, MA.
- LEGROS, F. (2006): "NDCs: A comparison of the French and German Point Systems," in Pension Reform: Issues and Prospects for Non-Financial Defined Contribution (NDC) Schemes, ed. by R. Holzmann, and E. Palmer, chap. 10, pp. 203–222. The World Bank, Washington, DC.
- MACCHERONI, C. (1998): "Frontiere della Sopravvivenza più Vicine? Alcuni Risultati Tratti dalla Recente Esperienza Italiana," *Diritto ed Economia dell'Assicurazione*, 40(4), 783– 805.
- MACCHERONI, C., AND T. BARUGOLA (2010): "E se l'aspettativa di vita continuasse la sua crescita? Alcune ipotesi per generazioni italiane 1950-2005," CeRP wp 97/10.
- MACCHERONI, C., AND I. LOCATELLI (1999): "I Grandi Anziani: Evoluzione del Fenomeno

in Italia alla Luce del Metodo delle Generazioni Estinte," *Rivista Italiana di Economia, Demografia e Statistica*, 53(1).

MEF (2009): "Mid-long Term Trends for the Pension, Health and Long-Term Care Systems," Ministry of Economy and Finance, Report n.11, Summary and Conclusions.

(2011): "Mid-long Term Trends for the Pension, Health and Long-Term Care Systems," Ministry of Economy and Finance, Report n.12, Summary and Conclusions.

- MLSP (2002a): "Gli Andamenti Finanziari del Sistema Pensionistico Obbligatorio," Ministry of Labour and Social Policies, Social Security Evaluation Committee (Italy).
 - (2002b): "Report on National Strategies for Future Pension Systems (Italy)," Ministry of Labour and Social Policies.
- MYERS, D., AND K. MANTON (1984): "Compression of Mortality, Myth or Reality?," *The gerontologist*, 24(4), 346–353.
- OLSHANSKI, S., B. CARNES, AND C. CASSEL (1993): "The Ageing of the Human Species," Scientific America, 268(4), 46–52.
- PALMER, E. (2006): "What is NDC?," in Pension Reform: Issues and Prospects for Non-Financial Defined Contribution (NDC) Schemes, ed. by R. Holzmann, and E. Palmer, chap. 2, pp. 17–33. The World Bank, Washington, DC.
- QUEISSER, M., AND E. R. WHITEHOUSE (2006): "Neutral or Fair?: Actuarial Concepts and Pension-System Design," OECD Social, Employment and Migration Working Papers n. 40.
- ROBINE, J.-M., E. CRIMMINS, S. HORIUCHI, AND Y. ZENG (2006): Human Longevity, Individual Life Duration, and the Growth of the Oldest-Old Population, vol. 4 of International Studies in Population. Springler.
- SAMUELSON, P. (1958): "An Exact Consumption-loan Model of Interest with or without the Social Contrivance of Money," *Journal of Political Economy*, 66.
- SORLIE, P., E. ROGOT, R. ANDERSON, N. JOHNSON, AND E. BACKLUND (1992): "Blackwhite mortality differences by family income," *Lancet*, 340(8815), 346–350.
- ZABA, B. (1979): "The Four Parameter Logit Life Table System," *Population Studies*, 33(1), 79–100.

Appendices

A Mortality forecasts: methodology

The construction of our life tables by cohort involves several steps. The first step consists in converting the indication of the experts on the human being limit characteristics (Robine, Crimmins, Horiuchi, and Zeng 2006) into an appropriate survival function. Following previous studies (see e.g. Duchêne and Wunsch 1993, Maccheroni 1998), we model the trend in endogenous mortality $q_{end}(x)$ by means of a Weibull model. It generates a process of mortality of an initial closed contingent of individuals of the same age according to the following survival function:

$$\ell(x) = exp\left[-\left(\frac{x-a}{m}\right)^b\right] \qquad x \ge a; a \ge 0; b, m > 0 \tag{A1}$$

The derivative of (A 1) provides the corresponding function of deaths which corresponds to the Weibull statistical distribution:

$$f(x) = \begin{cases} \frac{b}{m} \left(\frac{x-a}{m}\right)^{b-1} exp\left[-\left(\frac{x-a}{m}\right)^{b}\right] & x \ge a; a \ge 0; b, m > 0\\ 0 & \text{otherwise} \end{cases}$$

where we interpret f(x)dx as the fraction of components of an initial contingent that are eliminated at age [x, x + dx). The Weibull model, in particular the three-parameters type, permits fairly effective control of important features of the survival reference scenario, such as life expectancy at birth, the Lexis point and also the threshold below which mortality is considered avoidable. They are all functions of the parameters in equation (A 1) (Maccheroni 1998).

We then build the limit probabilities of dying due to exogenous or accidental causes $q_{eso}(x)$, i.e. accidents, traumas, etc. $q_{eso}(x)$ is valorised analyzing the trend of this set of causes of death for a group of developed countries,²³ thus constructing an initial table of minimum mortality by age. A fifth degree polynomial, which provides the analytical form of $q_{eso}(x)$, is then adapted to the empirical function obtained in this way.

The limit probabilities of dying $q_{lim}(x)(x=0,1,...)$ is obtained as the suitably perequated sum of (the discrete version of) the original components $q_{end}(x)$ and $q_{eso}(x)$. The abridged limit survival function is shown in table A 1. The synthetic characteristics of this mortality model are $e_0 = 109.4$ years and an extreme age of life $\omega = 125$.

Forecasts are then obtained as follows:

1. by establishing a time frame for the scenario provided by $q_{lim}(x)$ in relation to the recorded Italian mortality trends. A procedure based on the logit model (see Brass 1971, Zaba 1979) is used: the limit survival function $\ell_{lim}(x)$ obtained from $q_{lim}(x)$ is taken as standard and, using the observed survival functions (Maccheroni and Locatelli 1999) $\ell_{x,t}(t = 1945, 1946, 1998)$, the historical series of parameters a_t and b_t of the following relationship are studied:

$$Y_{x,t} = a_t + b_t Y_x^{lim} \tag{A2}$$

where Y_x^{lim} is the logit of the limit life table and $Y_{x,t}$ is the logit obtained from the observed $\ell_{x,t}$. Extrapolations are performed on the historical series of a_t and b_t in order to obtain new time sequences for parameters a_t^* and b_t^* (to establish if and when $a_t^* \to 0$ and $b_t^* \to 1$ when t diverges). In our case, these results are obtained when t = 2143 for females and when t = 2170 for males. On the basis of (A 2), it is then possible to obtain the sequence of projected life tables that reflect the characteristics of the limit situation in the period of time that stretches from 1999 to the two previously defined extremities of time;

²³ In details: Norway, Holland, Belgium, Spain, France, Finland, Italy, Great Britain, Sweden, Austria, Canada, USA, Australia, Japan, New Zealand.

Michele Belloni and Carlo Maccheroni

2. by linking the mortality models obtained as explained above with the on-going process of evolution of mortality. The gap between current and limit mortality is bridged by assuming a type of evolution that reflects the theory of expansion of mortality (Myers and Manton 1984). This step is carried out in two phases. In the first one, we define the evolution of mortality resulting from the most recent trends. In the age groups where the mortality trend is decreasing, the evolution of mortality is obtained by extrapolating the recent observed historical series of $q_{x,t}$ with a conventional exponential model. In the groups where mortality is growing, the evolution is first obtained by envisaging a stationary situation and subsequently by considering a tendentially decreasing evolution. In the second phase, we synthesize the projections for each year by averaging the results of the two mortality models (the logit A 2 and the exponential model). The average assigns gradually increasing linear weights to the life tables projected obtained from the logit (A 2) and decreasing linear weights to those obtained with the exponential model. The resulting projected life tables, therefore, initially take into account recent mortality trends and gradually assume the characteristics of the limit life table.

$Age_{\mathbf{x}}$	$\ell_{\mathbf{x}}$	$Age_{\mathbf{x}}$	$\ell_{\mathbf{x}}$	$Age_{\mathbf{x}}$	$\ell_{\mathbf{x}}$
0	100,000	40	99,831	85	98,985
1	99,997	45	99,799	90	98,242
5	99,992	50	99,768	95	$95,\!017$
10	99,983	55	99,735	100	85,236
15	99,968	60	$99,\!695$	105	67,726
20	99,948	65	$99,\!642$	110	42,180
25	99,923	70	99,566	115	15,306
30	99,894	75	99,454	120	1,980
35	99,863	80	99,280	125	40

Table A1. Limit life table: survivors from 100,000 live births

B conversion coefficients: formulas

Conversion coefficients in the B07 and B95 scenarios are given by equation (B1) while in the AB scenario they are given by equation (B2). Symbols are described in table B1.

28

Table B1. symbols

=

Symbol	Description	Value
x	retirement age	[57,65]
s(-s)	gender (gender of the widow(er))	m=male, f =female
co	cohort	[1945, 2000]
g_f	Expected long-run real GDP growth rate	0.015
$\ell_{x,s}/\ell_{x,s}^{co}$	Survivors of age x and gender s , according to cross- sectional/longitudinal mortality tables	our computations ^{a}
Ω/Ω^{co}	life span according to cross-sectional/cohort- and gender- specific mortality tables	our computations
$\ell_{x,s}^{ved}$	Probability for the age- x gender- s widow(er) to re-marry	INPS 1989
$\Theta_{s+t,s}$	Probability for the age- $x + t$ gender- s widow(er) to leave the family	INPS 1989
ϵ_s	Age difference between the (gender- s) pensioner and the widow(er)	+3(-3) if s=m(f)
Ψ	Quota of the pension revertible to the widow(er)	0.6
Φ_s	Reduction in the survivor's benefit due to earning test	0.9(0.7) if s=m(f)
k	Actuarial adjustment factor	0.4615^{b}

^a δ_x for co + x < 2010 are computed according to ISTAT 1990 mortality tables; δ_x for $2010 \le co + x \le 2013$ are computed according to ISTAT 2002 mortality tables. ^b law 247/2007

$$\delta_{x} = \left(\frac{\sum_{s=m,f} dir_{x,s} + ind_{x,s}}{2} - k\right)^{-1}$$
(B1)

$$dir_{x,s} = \sum_{t=0}^{\Omega-x} \frac{\ell_{x+t,s}}{\ell_{x,s}} (1+g_{f})^{-t}$$

$$ind_{x,s} = \Psi \Phi_{s} \sum_{t=0}^{\Omega-x} \frac{\ell_{x+t,s}}{\ell_{x,s}} (1 - \frac{\ell_{x+t+1,s}}{\ell_{x+t,s}}) (1+g_{f})^{-t} \Theta_{x+t,s}$$

$$\sum_{\tau=1}^{\Omega-x-t+\varepsilon_{s}} \frac{\ell_{x+t+\tau-\varepsilon_{s},-s}}{\ell_{x+t+1-\varepsilon_{s},-s}} (1 - \ell_{x+t+\tau-\varepsilon_{s},-s}) (1+g_{f})^{-\tau}$$

$$\delta_{x,s}^{co} = (dir_{x,s}^{co} + ind_{x,s}^{co})^{-1}$$
(B2)

$$dir_{x,s}^{co} = \sum_{t=0}^{\Omega^{co}-x} \frac{\ell_{x+t,s}^{co}}{\ell_{x,s}^{co}} (1+g_{f})^{-t}$$

$$ind_{x,s}^{co} = \Psi \Phi_{s} \sum_{t=0}^{\Omega^{co}-x} \frac{\ell_{x+t,s}^{co}}{\ell_{x,s}^{co}} (1 - \frac{\ell_{x+t+1,s}^{co}}{\ell_{x+t,s}^{co+\varepsilon_{s}}}) (1+g_{f})^{-t} \Theta_{x+t,s}$$

$$\sum_{\tau=1}^{\Omega^{co}-x-t+\varepsilon_{s}} \frac{\ell_{x+t+\tau-\varepsilon_{s},-s}^{co+\varepsilon_{s}}}{\ell_{x+t+1-\varepsilon_{s},-s}^{co+\varepsilon_{s}}} (1 - \ell_{x+t+\tau-\varepsilon_{s},-s}^{ved}) (1+g_{f})^{-\tau}$$

C Detailed results

In this appendix we report detailed simulation results for the base macroeconomic case $(r = 0.02, g_f = 0.015)$. Scenario B95 = law n. 335/1995; Scenario B07 = current legislation; Scenario AB = conversion coefficients are cohort- and gender-specific. DB = defined benefit; PR = *pro-rata*; DC = notional defined contribution. Indicators are in percentage points.

PVR	- MALE W	VHITE C.		Scenari	o B95					Scenaric	B07					Scenari	o AB		
rodo2	.t /scheme	60	5	retirem 62	ent age 63	64	ц С	60	5	retireme: 62	nt age 63	64	99 19	60	19	retireme 62	nt age 63	64	5
		8	5		8	5	8	8	5	3	8	5	8	8	5	3	8	5	8
1945 1946	DB	154.1 152.7	149.2 147.9	144.4 142.9	139.5 138.1	134.6 133.3	126.7 125.6	154.1 152.7	149.2 147.9	144.4 142.9	139.5 138.1	134.6 133.3	126.7 125.6	154.1 152.7	149.2 147.9	144.4 142.9	139.5 138.1	134.6 133.3	126.7 125.6
1947	DB	150.4	145.5	140.7	136.0	131.5	123.9	150.4	145.5	140.7	136.0	131.5	123.9	150.4	145.5	140.7	136.0	131.5	123.9
1948	DB	148.2	143.5	138.8	134.3	129.9	122.4	148.2	143.5	138.8	134.3	129.9	122.4	148.2	143.5	138.8	134.3	129.9	122.4
1949	a C	146.0	141.4 130.6	136.9	132.6	128.2	120.7	146.0 144.0	141.4 130.6	136.9	132.6	128.2	120.7	146.0 144.0	141.4 130.6	136.9	132.6 130.0	128.2	120.7
1951	DB	142.8	138.5	134.1	129.8	125.3	117.9	142.8	138.5	134.1	129.8	125.3	117.9	142.8	138.5	134.1	129.8	125.3	117.9
1952	DB	141.8	137.5	133.2	128.7	124.3	117.0	141.8	137.5	133.2	128.7	124.3	117.0	141.8	137.5	133.2	128.7	124.3	117.0
1953	DB	140.9	136.6	132.3	127.8	123.5	116.3	140.9	136.6	132.3	127.8	123.5	116.3	140.9	136.6	132.3	127.8	123.5	116.3
1954	н Ч	129.2	126.9	124.7	122.7	110.8	119.1	127.0	124.6	121.3	119.1	117.0	114.1	122.2	119.4	116.8	114.2	111.9	109.6
1956 1956	д н н	125.0	123.8	123.4	121.0	113.4	2.211	120.2	122.1	118.3	115.5 115.5	113.7 113.7	119.1	117.7	115.9	112 G	2.211	108.7	2.801
1957	PR.	124.2	122.4	120.7	113.8	112.2	110.8	120.7	118.7	115.9	114.2	112.6	110.1	115.5	113.2	111.1	109.0	107.1	105.3
1958	PR	122.6	121.0	114.2	112.6	111.2	109.9	119.0	116.2	114.5	113.0	110.5	109.1	113.3	111.2	109.3	107.4	105.6	104.0
1959	$_{\rm PR}$	121.1	114.4	112.9	111.4	110.2	109.0	116.3	114.7	113.2	110.8	109.4	108.2	111.2	109.3	107.5	105.8	104.2	102.6
1960	PR uu	114.4	112.9	111.6	110.3	109.2	108.2	114.7	113.3	110.9	109.6	108.4	106.3	109.2	107.4	105.8	104.2	102.7	101.3
1961	Р. Н.	112.7	111.4	110.2	109.1	108.1	107.3	113.0	110.7	109.5	108.4	106.3	105.3	107.1	105.5	104.0	102.5	101.2	99.9
1963	РК	109.4	109.9 108.4	107.6	106.8	106.1	105.5	110.4	2.601	105.7	104.9	104.1	104.4 102.3	102.8	101.6	100.4	99.3	99.7 98.2	93.0 97.2
1964	РК	107.7	106.9	106.2	105.6	105.0	104.6	107.0	105.1	104.3	103.6	101.9	101.3	100.7	99.66	98.5	97.6	96.6	95.8
1965	$_{\rm PR}$	106.0	105.4	104.8	104.4	104.0	99.5	104.2	103.5	102.9	101.3	100.8	100.3	98.5	97.6	96.7	95.9	95.1	94.3
1966	$_{\rm PR}$	104.3	103.9	103.5	103.2	98.8	98.5	102.5	102.0	100.4	100.0	9.60	98.1	96.4	95.6	94.9	94.2	93.5	92.9
1967	PR	102.6	102.4	102.1	98.0	97.7	97.5	100.8	99.3 5- 5	99.0	98.7	97.3	97.1	94.3	93.6	93.0	92.5	91.9	91.4
1060	ЧЧ	00.0	100.9	96.9 05.0	96.7 05 e	96.6 05 e	96.5 05 e	98.U	97.8 06.7	97.6 05.5	96.3 05 4	96.2 05 4	96.1 04 2	92.2 00 F	91.7	91.2	90.8 80 F	90.4 eo 1	89.98 0 0 0 0
1970	PR.	94.7	94.7	94.8	94.9	95.0	95.1	95.4	94.4	94.4	94.5	93.4	93.5	88.7	88.5	09.09 88.3	88.1	87.9	87.7
1971	DC	91.8	92.0	92.2	92.5	92.7	93.0	91.4	91.6	91.8	90.9	91.1	91.4	85.5	85.5	85.4	85.4	85.3	85.3
1972	DC	92.0	92.2	92.4	92.7	93.0	93.3	91.6	91.8	90.9	91.2	91.4	90.4	85.4	85.4	85.3	85.3	85.2	85.2
1973	D C L	92.2	92.4	92.7	93.0	93.3	93.6	91.9	91.0	91.2	91.4	90.5 20.5	90.7	85.4	85.3	85.3	85.2	85.2	85.1
1974	ט ני ק ב	92.5 92.8	92.7	93.0 93.3	93.3 93.6	93.6 93.9	93.9 90.2	91.4	91.3 91.6	91.5	90.6	90.8	91.0	85.3 85.3	85.3	85.2	85.2 85.2	85.1 85.1	85.1
1976	DC	93.1	93.3	93.6	93.9	90.3	90.6	91.7	90.8	91.0	91.2	90.3	90.6	85.3	85.3	85.2	85.2	85.1	85.1
1977	DC	93.4	93.6	93.9	90.5	90.7	90.9	90.9	91.1	91.3	90.5	90.7	90.9	85.3	85.3	85.2	85.2	85.1	85.1
1978	DC	93.7	93.9	90.6	90.8	91.0	91.2	91.2	91.4	90.6	90.8	91.0	90.1	85.3	85.2	85.2	85.1	85.1	85.1
1979	DC	93.9	90.6	90.8	91.0	91.2	91.5	91.5	90.6	90.8	91.0	90.2	90.4	85.3	85.2	85.2	85.1	85.1	85.0
1980		90.7	90.9	91.1	91.3 01.6	91.5	91.8	90.7	90.9	91.1	90.3	90.4	90.7	85.2 er 2	85.2 er o	85.1 ee 1	85.1 er 1	85.0 er o	85.0 85.0
1982		91.3	91.4	91.7	91.9	92.1 92.1	92.4	91.3	90.5	90.6	8.06 8.06	90.06	90.2	85.2	85.1	85.1	85.1	85.0	85.0
1983	DC	91.5	91.7	91.9	92.2	92.4	92.7	90.5	90.7	90.9	90.1	90.3	90.5	85.2	85.1	85.1	85.0	85.0	84.9
1984	DC	91.8	92.0	92.2	92.4	92.7	93.0	90.8	91.0	90.2	90.4	90.5	89.7	85.2	85.1	85.1	85.0	85.0	84.9
1985	D C D	92.0	92.2	92.5 02.7	92.7	93.0	89.6	91.1	90.3	90.4	90.6	89.8	90.0	85.1	85.1	85.0	85.0	84.9	84.9
1987		92.3 02.5	0.28 0.28	92.7	93.U 89.8	60.0	89.9 90.2	90.4 90.6	90.5 90.8	90.7	89.9 90.2	90.1 90.4	80.5 80.5	85.1 85.1	85.1 85.1	85.0	85.0	84.9 84 0	84.9 84.0
1988	DC	92.8	93.0	89.9	90.1	90.3	90.5	90.9	90.1	90.3	90.4	89.6	89.8	85.1	85.0	85.0	84.9	84.9	84.8
1989	DC	93.0	90.0	90.2	90.4	90.6	90.8	90.2	90.4	90.5	89.7	89.9	90.1	85.1	85.0	85.0	84.9	84.9	84.8
1990	DC	90.1	90.3	90.5	90.6	90.8	91.1	90.4	90.6	89.8	90.0	90.2	89.4	85.1	85.0	85.0	84.9	84.9	84.8
1991		90.4	90.5	90.7	90.9	91.1	91.3 01 6	90.7	89.9	90.1 00.2	90.3	89.5	89.6	85.0 85.0	85.0 85.0	84.9	84.9	84.8	84.8
1993		9.06	0.16	0.1.6	91.4	91.6	6.16	90.3	20.6 90.4	2.06	8.68	0.06	89.2	85.0	84.9	84.9	84.8	84.8	84.7
1994	DC	91.1	91.3	91.5	91.7	91.9	92.1	90.5	89.8	89.9	90.1	89.3	89.4	85.0	84.9	84.9	84.8	84.8	84.7
1995	DC	91.3	91.5	91.7	91.9	92.2	89.1	89.9	90.0	90.2	89.4	89.5	89.7	85.0	84.9	84.9	84.8	84.7	84.7
1996	DC	91.6	91.8	92.0	92.2	89.2	89.3	90.1	90.3	89.5	89.7	89.8	89.0	84.9	84.9	84.8	84.8	84.7	84.6
1997		91.8 02.1	92.0 92.3	92.2 80.4	89.3 80.3	89.4 80.7	89.6 80.8	90.4 80.7	89.6 90.0	89.8	89.9 40.7	89.1 80.4	89.3 80.3	84.9 84.9	84.9 84 8	84.8 84.8	84.7	84.7 9.4.6	84.6 84.6
1000		92.1 92.3	80.5 2	80.7	0.00 8 08	80.0	09.00 90.1	90.0	о <u></u>	90.0 89.4	2.60 2.08	80.6	0 9 0 8 8 0 8 8 0	84.9	04.0 84.8	04.0 84.7	84.7	0.4.0 84.6	84.5
2000	DO	89.6	89.7		90.0	90.1	90.3	90.2	89.4	89.6	89.7	88.9	89.0	84.8	84.8	84.7	84.6	84.6	84.5

PVR	- MALE B.	LUE C.		Scenar.	io B95					Scenaric	B07					Scenarie	o AB		
				retirem	ent age					retireme	nt age					retireme	nt age		
cohor	t/scheme	60	61	62	63	64	65	60	61	62	63	64	65	60	61	62	63	64	65
1945	DB	137.4	132.6	124.7	117.0	109.8	103.0	137.4	132.6	124.7	117.0	109.8	103.0	137.4	132.6	124.7	117.0	109.8	103.0
1946	DB	136.7	131.9	123.8	116.2	109.0	102.3	136.7	131.9	123.8	116.2	109.0	102.3	136.7	131.9	123.8	116.2	109.0	102.3
1947 1948	DB aC	135.0	130.9	122.9	115.4	108.3 107.8	101.7	136.0	130.9	122.9	115.4	108.3	101.7	136.0	130.9	122.9	115.4	108.3 107 s	101.7
1949	DB	134.2	129.3	121.5	114.2	107.2	100.6	134.2	129.3	121.5	114.2	107.2	100.6	134.2	129.3	121.5	114.2	107.2	100.6
1950	DB	133.1	128.4	120.6	113.3	106.4	99.9	133.1	128.4	120.6	113.3	106.4	99.9	133.1	128.4	120.6	113.3	106.4	99.9
1951	DB ar	132.1	127.4	119.7	112.4 111.6	105.6 104 9	99.2 98.6	132.1	127.4	119.7	112.4 111.6	105.6 104 9	99.2 98.6	132.1	127.4	118.8	112.4 111.6	105.6 104 9	99.2 98.6
1953	DB	130.1	125.5	117.9	110.9	104.2	98.0	130.1	125.5	117.9	110.9	104.2	98.0	130.1	125.5	117.9	110.9	104.2	98.0
1954	DB	129.2	124.7	117.2	110.2	103.7	97.5	129.2	124.7	117.2	110.2	103.7	97.5	129.2	124.7	117.2	110.2	103.7	97.5
1955	DB	128.6	124.1	116.8	109.8	103.3	97.2	128.6	124.1	116.8	109.8	103.3	97.2	128.6	124.1	116.8	109.8	103.3	97.2
1956	DB	128.1	115.6	116.3	109.4	103.0	96.9 102.0	128.1	123.6 112 E	116.3 100.6	109.4	103.0	96.9 103 3	128.1	123.6	116.3	109.4	103.0	96.9 00 1
1958	PR PR	116.8	114.9	108.5	106.7	105.0	103.5	113.6	110.7	108.8	107.0	104.4	102.8	108.6	106.3	104.1	102.1	1.101	98.3
1959	$_{\rm PR}$	115.8	109.5	107.7	106.1	104.6	103.1	111.7	109.8	108.0	105.5	103.9	102.5	107.2	105.0	103.0	101.1	99.2	97.5
1960	PR	110.3	108.6	107.0	105.5	104.1	102.8	110.6	108.9	106.4	104.8	103.4	101.1	105.7	103.7	101.8	100.0	98.3	96.7
1961	РК	109.3	107.7	106.2	104.9	103.6	102.5	109.6	107.1	105.6	104.2	102.0	100.7	104.3	102.4	100.7	99.0	97.4	95.9
1962 1963	PR.	107.3	106.0	104.8 104.8	104.3 103.7	103.2	101.9	106.6	105.3	104.9 103.2	102.0	0.101 0.101	100.4 99.0	101.4	1.101	99.5 98.4	98.0 97.0	96.5 95.6	95.2 94.4
1964	РК	106.2	105.1	104.1	103.2	102.3	101.6	105.6	103.5	102.4	101.4	99.5	98.6	99.9	98.5	97.2	95.9	94.7	93.6
1965	$_{\rm PR}$	105.3	104.3	103.5	102.7	101.9	97.5	103.7	102.7	101.7	99.8	0.06	98.2	98.5	97.3	96.1	94.9	93.8	92.8
1966	РК	104.3	103.5	102.8	102.1	97.8	97.2	102.7	101.8	100.0	99.2	98.5	96.8	97.1	96.0	94.9	93.9	92.9	92.0
1967	PR ac	103.4	102.7	102.2	98.0	97.4 06.0	96.8 96.8	101.7	100.0	99.3 08.6	98.6 07 0	97.0 92.7	96.4 06.1	95.7 01.1	94.7 02.7	93.8 00 7	92.9 01.0	92.0 61.1	91.2
1969	д Ц Ц	101 6	0.201	97.9 07.9	91.4 96.8	90.9 96.4	90.0 96 1	0.89 8.80	99.1 08.3	90.05 96.8	97.U 96.4	96.0 96.0	90.L 94.6	94.4 93.0	93.0 92.2	91.5	91.9 00 0	5 Ub	80.6
1970	PR	97.1	96.7	96.4	96.2	95.9	95.7	97.8	96.4	96.1	95.8	94.4	94.2	91.5	91.0	90.4	89.8	89.3	8.88
1971	$_{\rm PR}$	96.1	95.9	95.7	95.6	95.4	95.4	95.7	95.5	95.3	94.1	93.9	93.8	90.1	89.7	89.2	88.8	88.4	88.0
1972	PR rr	95.2	95.1	95.0	95.0	95.0	95.0	94.8	94.7	93.6 22.0	93.5	93.5	92.3 01.0	88.8	88.5	88.1	87.8	87.5	87.2
1074	Ч Ч С С	94.2	94.3 01.6	94.3 01 s	94.4 02 1	0.94.0 0.3	94.7 02.6	93.9	87.8	92.8	92.9 80.4	91.8 80.6	8.16 8.08	81.4	81.2	0.78	20.7	0.08 0.19	80.3 82.0
1975	DC	91.6	91.8	92.0	92.3	92.6	88.9	90.2	90.4	89.5	£.66 89.7	89.8	88.9	84.2	84.2	84.1	84.0	83.9	83.8
1976	DC	91.8	92.1	92.3	92.5	89.0	89.2	90.4	89.6	89.7	89.9	89.0	89.2	84.2	84.1	84.0	83.9	83.8	83.8
1977	DC	92.1	92.3	92.6	89.1	89.3	89.5	89.7	89.8	90.0	89.1	89.3	89.5	84.1	84.1	84.0	83.9	83.8	83.7
1978		92.4	92.6 80.4	89.3 80.6	89.4 80.7	89.68 80.0	89.8	90.0 90.3	90.1 80.4	89.3 80.6	89.4 80.7	89.6 88.8	88.7	84.1 84.1	84.1 84.1	84.0	83.9 83.9	x x x x	83.7
1980	DC	89.6	89.7	89.9	90.0	90.2	90.4	89.6	89.7	89.9	89.0	89.1	89.3	84.1	84.1	84.0	83.9	83.8	83.7
1981	DC	89.8	90.0	90.1	90.3	90.5	90.7	89.8	0.06	89.1	89.3	89.4	88.5	84.1	84.0	84.0	83.9	83.8	83.7
1982		90.1	90.2	90.4	90.6	90.8	91.0	90.1 80.4	89.2	89.4	89.5 66 6	88.7	80.00 80.00 70	84.1	84.0	83.9	80.00 80.00 80.00	x x x	83.7
1963		90.6	5.06 20.7	0.06 90.9	91.1 91.1	91.3	91.6	89.6	89.7	09.00 88.9	89.1	89.2	09.1 88.3	84.0	84.0	83.9	83.8	83.7	83.6
1985	DC	90.8	91.0	91.2	91.4	91.6	88.3	89.9	89.1	89.2	89.3	88.5	88.6	84.0	83.9	83.9	83.8	83.7	83.6
1986	DC	91.1	91.2	91.5	91.7	88.4	88.6	89.2 55 ·	89.3 22.2	89.4	88.6	88.8	88.9	84.0	83.9	83.8	83.8	83.7	83.6
1988		91.6 91.6	91.5 8.19	88.7	88.0 8.88	89.0	80.1 80.1	89.4	89.68 88.9	89.0	80.1 80.1	88.3 88.3	88.4	84.0	83.9 83.9	83.8 83.8	83.7	83.6	83.6 83.6
1989	DC	91.8	88.88	89.0	89.1	89.2	89.4	89.0	89.1	89.3	88.5	88.6	88.7	84.0	83.9	83.8	83.7	83.6	83.5
1990	DC	89.0	89.1	89.2	89.3	89.5	89.7	89.3	89.4	88.6	88.7	88.8	88.0	83.9	83.9	83.8	83.7	83.6	83.5
1991	DC	89.2	89.3	89.5	89.6	89.8	90.0	89.5	88.7	88.8	89.0	88.2	88.3	83.9	83.8	83.8	83.7	83.6	83.5
1992	DC	89.4	89.6	89.7	89.9	90.0	90.2 20.7	x . x . x	89.0	89.1	88.3 2 1	88.4	88.5 0 1	83.9	83. 8 8 0	83.7	83.7	83.6	83.5 7
1994		89.9	0.06 0.00	90.0 90.2	90.1 90.4	90.6 90.6	90.8 80.8	89.3	88.6	88.7	00.00 8.88	88.0	07.0 88.1	83.9	83.8 83.8	83.7	83.6	83.5	83.5
1995	DC	90.1	90.3	90.5	90.6	90.8	87.8	88.7	88.8	88.9	88.1	88.3	88.4	83.8	83.8	83.7	83.6	83.5	83.4
1996	DC	90.4	90.5	90.7	90.9	87.9	88.0	88.9	89.0	88.3	88.4	88.5	87.7	83.8	83.7	83.7	83.6	83.5	83.4
1997	DC	90.6	90.8	91.0	88.1	88.2	88.3	89.2 22	88.4	88.5	88.7	87.9	88.0	83.8	83.7	83.6	83.6	83.5	83.4
1000	ב כ ב	90.9 01 1	91.0	X X X X X X X	88.3 88.6	88.4 88.7	8 8 8 8 8 8	8 8 8 0	88 0	x x x x x x	0.88 88.3	88.1 88.1	8 2 . Z	x 2, 2 x 2, 2 x 2, 2	83.7	83.0 83.6	83.5 83.5	83.5 83.4	83.4 83.4
2000	DC	88.4	88.5	88.7	88.8 88.8	88.9	89.1	89.0	88.3	88.4	88.5	87.7	87.8	83.8	83.7	83.6	83.5	83.4	83.3

PVR	- FEMALE	WHITE	v	Scenari	o B95					Scenaric	B07					Scenari	o AB		
				retirem	ant age					retireme	nt age					retireme	nt age		
cohor	t/scheme	60	61	62	63	64	65	60	61	62	63	64	65	60	61	62	63	64	65
945	DB	151.0	146.9	142.7	138.2	133.7	126.1	151.0	146.9	142.7	138.2	133.7	126.1	151.0	146.9	142.7	138.2	133.7	126.1
946	DB	150.3	146.2	141.7	137.3	132.9	125.5	150.3	146.2	141.7	137.3	132.9	125.5	150.3	146.2	141.7	137.3	132.9	125.5
947	DB	149.3	144.9	140.5	136.3	132.0	124.7	149.3	144.9	140.5	136.3	132.0	124.7	149.3	144.9	140.5	136.3	132.0	124.7
948	DB c	147.9	143.6	139.4	135.3	131.1	123.8	147.9	143.6	139.4	135.3	131.1	123.8	147.9	143.6	139.4	135.3	131.1	123.8
.949 050		146.6 145.3	142.4 141 3	138.3 137.3	134.3	130.1	122.9	145.6 145.3	142.4 141 3	138.3	134.3	130.1	122.9	146.6 145.3	142.4 141 3	138.3	134.3	130.1	122.9
951	DB	144.4	140.5	136.5	132.5	128.5	121.4	144.4	140.5	136.5	132.5	128.5	121.4	144.4	140.5	136.5	132.5	128.5	121.4
952	DB	143.8	139.8	135.9	131.9	128.0	121.0	143.8	139.8	135.9	131.9	128.0	121.0	143.8	139.8	135.9	131.9	128.0	121.0
.953	DB	143.0	139.2	135.3	131.4	127.5	120.6	143.0	139.2	135.3	131.4	127.5	120.6	143.0	139.2	135.3	131.4	127.5	120.6
1954	\mathbf{PR}	128.8	126.7	124.8	123.0	121.4	120.0	126.6	124.4	121.4	119.5	117.7	115.0	120.0	117.4	114.9	112.6	110.4	108.2
1955	$_{\rm PR}$	127.3	125.4	123.6	122.0	120.6	113.8	125.0	122.0	120.1	118.3	115.7	114.1	117.8	115.4	113.1	110.9	108.8	106.8
1956	РК	125.8	124.0	122.5	121.0	114.4	113.0	122.5	120.6	118.9	116.3	114.7	113.3	115.7	113.4	111.3	109.2	107.3	105.4
1957	РК	124.3	122.8	121.4	114.8	113.4	112.1	120.9	119.2	116.6	115.1	113.8	111.4	113.6	111.5	109.5	107.6	105.8	104.1
1958	Р. П.	122.9	121.5	115.1	113.7	112.5	111.3	119.3	116.8 115 5	115.4	114.1	111.8	110.6	111.5	109.5 107 e	107.7	105.9	104.3	102.7
1060	100	0.141	11011	0.011	0.211	0.111	0.011	115.2	1111	711 0	0.211	0.011	0.20T	107.2	108 7	104.1	0.4.01	101 2	0.00
1961	PR	113.4	112.4	111.4	110.5	109.7	109.0	113.8	111.7	110.7	109.8	107.9	107.1	105.2	103.7	102.3	101.0	2.66	38.5
1962	PR	112.0	111.1	110.3	109.6	108.9	108.4	111.3	110.4	109.6	107.7	107.0	106.4	103.2	101.9	100.6	99.4	98.3	97.2
1963	$_{\rm PR}$	110.6	109.9	109.2	108.6	108.1	107.7	109.9	109.1	107.4	106.7	106.1	104.5	101.2	100.0	98.9	97.9	96.8	95.9
1964	\mathbf{PR}	109.2	108.7	108.1	107.7	107.4	107.1	108.5	106.8	106.3	105.8	104.2	103.7	99.3	98.2	97.2	96.3	95.4	94.6
1965	$_{\rm PR}$	107.9	107.5	107.1	106.8	106.6	102.3	106.1	105.6	105.2	103.7	103.3	103.1	97.3	96.4	95.6	94.8	94.0	93.3
1966	PR	106.6	106.3	106.1	106.0	101.7	101.6	104.8	104.4	103.0	102.7	102.5	101.2	95.4	94.6	93.9	93.2	92.6	91.9
1967	PR uu	105.3	105.2	105.1	101.0	100.9	100.9	103.4	102.1	101.9	101.8	100.5	100.5	93.5 01 r	92.8	92.2	91.7	91.1	90.6
1060	Р.К. Д	104.1	104.1	100.1	100.1	100.1	100.2	1001	100.9	100.9	99.7 00 2	99.7 00 2	99.8 0 e o	91.5	91.1	90.6 80.3	90.1 80.0	89.7	89.3 8 9 9
1970	PR.	98.4	98.6	0.86 8.86	0.99.0	99.3	99.66 99.6	1.001	100.1 98.2	98.4	98.6 98.6	97.6	9.7.8	90.0 88.5	88.2	88.0	87.8	87.6	87.4
1971	DC	95.2	95.6	96.0	96.4	96.8	97.3	94.9	95.2	95.6	94.8	95.2	95.6	85.0	84.9	84.8	84.8	84.7	84.7
1972	DC	95.5	95.8	96.3	96.7	97.1	97.6	95.1	95.5	94.7	95.1	95.5	94.6	84.8	84.8	84.7	84.7	84.6	84.5
1973	DC	95.8	96.2	96.6	97.0	97.5	98.0	95.4	94.6	95.0	95.4	94.6	95.0	84.8	84.7	84.6	84.6	84.5	84.5
1974	DC	96.1	96.5	96.9	97.4	97.9	98.4	94.6	95.0	95.4	94.6	95.0	95.4	84.7	84.7	84.6	84.5	84.5	84.4
1975	DC	96.5	96.9	97.3	97.8	98.3	94.6	95.0	95.4	94.6	95.0	95.4	94.6	84.7	84.7	84.6	84.5	84.5	84.4
1976	D C C	96.8 07 0	97.3	97.7	98.2	94.6 67.0	95.0	95.4 04 7	94.7	95.0 or 1	95.4 04.7	94.6 07.0	95.0	84.7	84.7	84.6	84.5	84.5	84.4
1.020T	D C D	97.2	97.79	1.26	94.7	95.0	95.4	94.7	95.0	95.4	94.7	95.U	95.4	84.7	84.0	84.6	84.5	84.5 7	84.4
1070		97.0 07.0	98.0	94.7 05.0	90.1 05.4	90.4 05.8	90.9 06.2	90.U 05.2	90.4 04.7	94.7 05.0	95.1 05.4	95.4 04.7	94.7 05 1	84.7 81.6	84.0 84.6	0.4.0 7.1.8	0.4.0 7.7.8	0.45 84.0	84.4
1980	DC	94.7	95.0	95.4	95.8	96.2	90.2 96.6	94.7	95.0	95.4	94.7	95.1	95.5	84.6	84.5	84.5	84.4	84.4	84.3
1981	DC	95.0	95.4	95.7	96.1	96.6	97.0	95.0	95.4	94.7	95.0	95.4	94.7	84.6	84.5	84.5	84.4	84.4	84.3
1982	DC	95.3	95.7	96.1	96.5	97.0	97.4	95.3	94.7	95.0	95.4	94.7	95.1	84.6	84.5	84.4	84.4	84.3	84.3
1983	DC	95.7	96.1	96.5	96.9	97.4	97.8	94.7	95.0	95.4	94.7	95.1	95.5	84.5	84.5	84.4	84.4	84.3	84.2
1984	DC	96.0	96.4	96.8	97.3	97.8	98.3	95.0	95.4	94.7	95.1	95.5	94.8	84.5	84.4	84.4	84.3	84.3	84.2
1985		96.4 06.7	96.8	97.2	97.7 08.0	98.1	94.8 01 0	95.3	94.7	95.1 of 4	95.5 04 e	94.8	95.2 05.2	84.5	84.4	84.4	84.3	84.2	84.2
1987		0.76	1.16	6.76	94.8	95.2	92.6	95.0	95.4	94.8	95.1	95.5	94.9	84.4	84.4	84.3	84.2	84.2	84.1
1988	DC	97.4	97.8	94.8	95.2	95.6	96.0	95.4	94.8	95.1	95.5	94.9	95.3	84.4	84.3	84.3	84.2	84.2	84.1
1989	DC	97.7	94.8	95.2	95.5	96.0	96.4	94.8	95.1	95.5	94.9	95.3	95.7	84.4	84.3	84.3	84.2	84.1	84.1
1990	DC	94.8	95.1	95.5	95.9	96.3	96.8	95.1	95.5	94.9	95.2	95.6	95.0	84.3	84.3	84.2	84.2	84.1	84.1
1991	DC	95.1	95.5	95.9	96.3	96.7	97.2	95.4	94.9	95.2	95.6	95.0	95.4	84.3	84.3	84.2	84.1	84.1	84.0
1992	DC	95.5	95.8	96.2	96.7	97.1	97.6	94.8	95.2	95.6	95.0	95.4	95.8	84.3	84.2	84.2	84.1	84.1	84.0
1004		95.8 06 1	96.2 0 <i>e e</i>	96.6	97.1	97.5	98.0	95.2 05 5	95.6 05.0	95.U	95.4 05.7	95.8 05 1	95.I	84.3	84.2	84.1	84.1	84.0	84.0
1995	DC	96.5	0.0 <i>e</i>	97.4	#.1 <i>6</i> 8.78	94.3 98.3	95.2	94.9	95.3	95.7	95.1	95.5	95.9	84.2	84.1	84.1	84.0	84.0	83.9
1996	DC	96.8	97.3	97.7	98.2	95.2	95.6	95.3	95.7	95.1	95.5	95.9	95.3	84.2	84.1	84.1	84.0	83.9	83.9
1997	DC	97.2	97.6	98.1	95.2	95.6	96.0	95.6	95.1	95.5	95.9	95.3	95.7	84.1	84.1	84.0	84.0	83.9	83.9
1998	DC	97.5	98.0	95.2	95.6	96.0	96.5	95.1	95.4	95.8	95.3	95.7	96.1	84.1	84.0	84.0	83.9	83.9	83.8
1999		97.9	95.2 05.2	95.6 05.0	96.0	96.4 06.8	96.9 07 -	95.4 of e	95.8 05.3	95.3 05.6	95.7 06.0	96.1 of f	95.5 0 r 0	84.1	84.0	84.0	83.9	83.8	83.8
2000	DC	90.2	0.08	80.8	90.4	80.8	91.3	90.8	90.3	9.06	90.0	90.0	90.9	84.U	84.U	83.9	83.9	83.8	83.8

		2	retirem	ent age					retireme	nt age								
ç	en.		63	9 8	Ч	ц Ц	en	61	63	5	БЛ	н Ц	e O	19	retireme	ent age 63	61	u u
2	6	3		8	5	B	B	5	8	B	5			5	9	8	5	
	126.5 126.0	121.9 121.4	117.4 116.6	109.8 109.2	102.8 102.2	96.1 95.6	126.5 126.0	121.9 121.4	117.4 116.6	109.8 109.2	102.8 102.2	96.1 95.6	126.5 126.0	121.9 121.4	117.4 116.6	109.8 109.2	102.8 102.2	96.1 95.6
	125.6	120.7	116.0	108.6	101.7	95.3	125.6	120.7	116.0	108.6	101.7	95.3	125.6	120.7	116.0	108.6	101.7	95.3
	124.8	120.0	115.4	108.1	101.3	94.8	124.8	120.0	115.4	108.1	101.3	94.8	124.8	120.0	115.4	108.1	101.3	94.8
	123.7 122.6	118.1	114.5 113.6	106.4	100.5 99.7	94.1 93.3	123.7 122.6	1181	114.5 113.6	106.4	100.5 99.7	94.1 03.3	123.7	119.1	114.5 113.6	107.4	100.5 99.7	94.1 03 3
	121.7	117.3	112.8	105.7	0.00	92.8	121.7	117.3	112.8	105.7	99.0	92.8	121.7	117.3	112.8	105.7	99.0	92.8
	120.9	116.4	112.0	105.0	98.4	92.2	120.9	116.4	112.0	105.0	98.4	92.2	120.9	116.4	112.0	105.0	98.4	92.2
	120.1	115.6	111.3	104.4	97.9	91.8	120.1	115.6	111.3	104.4	97.9	91.8	120.1	115.6	111.3	104.4	97.9	91.8
	119.6	115.2	110.9	104.0	97.6 07.0	91.5	119.6	115.2	110.9	104.0	97.6 07.0	91.5	119.6	115.2	110.9	104.0	97.6	91.5
	119.1	0.110.6	0.011	107.2	97.3 101.0	91.3	1.00.6	107 F	105.6	103.0	97.3 101.2	91.3 00 8	103.6	01 2	0.011	103.7 06.7	97.0	91.3 0.0
	111.9	110.2	108.6	102.4	100.9	99.5	108.9	107.0	104.4	102.7	101.2	98.8	102.4	100.1	98.0	96.0	94.1	92.3
	111.4	109.8	103.7	102.2	100.8	99.5	108.3	105.7	104.0	102.5	100.2	98.9	101.3	99.1	97.1	95.3	93.5	91.8
	111.0	104.9	103.4	102.0	100.8	99.6	106.8	105.2	103.7	101.4	100.1	99.0	100.1	98.1	96.3	94.5	92.9	91.3
	105.9	104.5	103.1	101.9	100.8	99.8	106.2	104.8	102.5	101.3	100.1	98.0	0.06	97.2	95.5	93.8	92.3	90.9
	105.4	104.1	102.9	101.8	100.8	99.9	105.7	103.5	102.3	101.1	99.1	98.2	97.9	96.3	94.7	93.2	91.8	90.4
	104.8	103.7	102.7	101.7	100.9	100.1	104.2	103.1	102.0	100.0	99.1 00.1	98.3	96.9 97 9	95.3	93.9	92.5	91.2	90.0
	104.0	103.1	102.4	2 101	101 1	100.6	103.7	101 5	100 7	99.99 000	1.99.1 0.8 0	91.4 07.6	90.0 94.8	94.4 03.6	93.1	81.8 01.9	0.06	0.08 0.08
	103.6	1.02.9	102.3	101.8	101.3	0.001	102.0	101.2	100.5	0.80	98.3	8.76	03.9	92.7	5.16	2.16	2.06	- 88
	103.3	102.8	102.3	101.9	2.701	97.3	101.6	101.0	99.4	98.9	98.4	96.9	92.9	91.9	91.0	90.1	89.2	88.4
	103.0	102.6	102.3	98.2	97.9	97.6	101.3	99.8	99.3	0.06	97.5	97.2	92.0	91.2	90.3	89.6	88.8	88.1
	102.8	102.6	98.6	98.3	98.1	97.9	100.0	99.66	99.3	97.9	97.7	97.5	91.1	90.4	89.7	89.0	88.4	87.7
	102.6	98.7	98.6 0.00	98.4 06.6	98.3	98.2	99.7	99.5 25.2	98.2	98.0	97.9	96.7	90.3	89.6	89.1	88.5	87.9	87.4
	90.7 08 3	0.00 0.80	98.0 08.4	93.0 98.7	93.0 98.6	90.0 98.0	99.4 07 0	90.0 0 8 0	98.2	98.2 06.0	97.0	97.0	88.4 88.4	00.90 0.88	000.0 87.6	87.9	0.70 86.0	86.6
	97.8	98.0	98.2	98.4	98.7	0.06 0.06	97.5	97.6	96.7	96.8	0.16	96.0	87.3	87.0	86.8	86.5	86.3	86.0
	94.9	95.2	95.6	96.0	96.4	96.8	94.5	93.7	94.1	94.4	93.6	93.9	84.0	83.9	83.8	83.7	83.6	83.5
	95.1	95.5	95.8	96.2	96.7	97.1	93.7	94.0	94.3	93.5	93.8	94.2	83.8	83.7	83.7	83.6	83.5	83.4
	95.4	95.8	96.1	96.6	97.0	93.3	93.9	94.3	93.5	93.8	94.1	93.3	83.8	83.7	83.6	83.5	83.4	83.3
	95.7	96.1 22 I	96.5 22.2	96.9	93.4	93.7	94.3	93.5	93.8	94.2	93.4	93.7	83.7	83.6	83.5	83.4	83.3	83.2
	96.1 06 r	96.5 06.0	96.9	93.4 02.6	93.8	94.1	93.6	93.9	94.2	93.4 02.6	93.8	94.1	83.7	83.6	83.5 83.5	83.4	50 50 50 50 50 50 50 50 50 50 50 50 50 5	27.52
	96.8	90.9 93.6	0.08	94.2	94.6 94.6	95.0	90.9 94.3	94.0 93.6	93.9	94.2	94.2	4.06 8.86	83.7	83.6	83.5	83.4	83.3	83.2
	93.6	93.9	94.3	94.6	95.0	95.4	93.6	93.9	94.3	93.5	93.8	94.2	83.7	83.6	83.5	83.4	83.3	83.2
	93.9	94.3	94.6	95.0	95.3	95.7	93.9	94.3	93.5	93.9	94.2	93.4	83.6	83.6	83.5	83.4	83.3	83.2
	94.3	94.6	94.9	95.3	95.7	96.1	94.3	93.6	93.9	94.2	93.5	93.8	83.6	83.5	83.4	83.3	83.2	83.1
	94.6	94.9 05 2	95.3 05.3	95.7 06.1	96.1 06 E	96.5 06.0	93.6	93.9	94.2 03 6	93.5	93.9	94.2 09 E	83.6 83.6	83.5 69 5	83.4	83.3	83.2	83.1
	95.3	95.6	96.0	96.4	6.96	93.5	94.3	93.6	93.9	94.3	93.5	93.9	83.5	83.4	83.3	83.2	83.1	83.1
	95.6	96.0	96.4	96.8	93.6	93.9	93.6	94.0	94.3	93.6	93.9	94.3	83.5	83.4	83.3	83.2	83.1	83.0
	96.0	96.4	96.8	93.6	94.0	94.3	94.0	94.3	93.6	94.0	94.3	93.6	83.5	83.4	83.3	83.2	83.1	83.0
	96.3	96.7	93.7	94.0	94.3	94.7	94.3	93.7	94.0	94.3	93.6	94.0	83.4	83.4	83.3	83.2	83.1	83.0
	96.6	93.7	94.0	94.4	94.7	95.1 07 7	93.7	94.0	94.3	93.7	94.0	94.4	83.4	83.3	83.2	83.1	83.0	82.9
	1.06	94.0	94.7	94.7 95 1	90. F	0.00 05 0	94.4	94.4 03.8	1.06	94.0	94.4 03.8	1.06	83.4	00.00 8.00 8.00	83.2	83.1	83.0	8.00
	94.4	94.7	95.1	95.5	95.9	96.3	93.8	94.1	94.4	93.8	94.1	94.5	83.3	83.2	83.2	83.1	83.0	82.9
	94.7	95.1	95.5	95.8	96.3	96.7	94.1	94.4	93.8	94.2	94.5	93.9	83.3	83.2	83.1	83.0	82.9	82.8
	95.1	95.4	95.8	96.2	96.6	97.1	94.5	93.9	94.2	94.5	93.9	94.2	83.3	83.2	83.1	83.0	82.9	82.8
	95.4	95.8	96.2	96.6	97.0	94.0	93.9	94.2	94.5	93.9	94.3	94.6	83.3	83.2	83.1	83.0	82.9	82.8
	95.8	96.1	96.5 2 2 2	97.0	94.0	94.4	94.2	94.6	94.0	94.3	94.7	94.0	83.2	83.1	83.0	82.9	82.8	82.7
	96.1 96.4	96.5 06 0	96.9 04 1	94.U 94.J	94.4 04.8	94.8 05.9	94.6 04.0	94.0 04 3	94.3 04.7	94.7 04.1	94.1 04 4	94.4 04 s	83.2	83.1	83.0	82.9 8 2 0	x x x x x x	2.2.8
	8.96	94.1	94.4	94.8	95.2	95.6	94.4	94.7	94.1	94.5	94.8	94.2	83.1	83.0	82.9	82.9	82.8	82.7
	0.01	94.4	94.8	95.2	95.5	96.0	2 7 0	1 10	10	0.10	c 10	04 6	831	0 00				80.6

TAX	- MALE W	HITE C		Scena	rio B95					Scenari	o B07					Scenar	io AB		
				retirem	ent age				-	retireme	nt age					retirem	ent age		
cohor	t/scheme	60	61	62	63	64	65	60	61	62	63	64	65	60	61	62	63	64	65
1945	DB	37.5	40.5	46.3	48.8	86.6	86.8	37.5	40.5	46.3	48.8	86.6	86.8	37.5	40.5	46.3	48.8	86.6	86.8
1946	DB DB	37.7	43.3	45.8	48.3	86.1 er r	86.4	37.7	43.3	45.8	48.3	86.1 85 E	86.4	37.7	43.3	45.8	48.3	86.1 er r	86.4
1947 1948	DB	40.4 39.8	42.1	40.1 44.5	46.9	86.2	87.1	40.4 39.8	42.1 42.1	40.1 44.5	41.0 46.9	86.2	87.1	40.4 39.8	42.1	40.1 44.5	47.0 46.9	86.2	87.1
1949	DB	39.2	41.4	43.8	47.8	87.7	89.8	39.2	41.4	43.8	47.8	87.7	89.8	39.2	41.4	43.8	47.8	87.7	89.8
1950	DB	38.6	40.9	45.0	50.6	89.1	89.0	38.6	40.9	45.0	50.6	89.1	89.0	38.6	40.9	45.0	50.6	89.1	89.0
1951	au au	30.2 2 2 2 2	42.0	45.6 47.6	51.1	x x x x x	88.6 88.0	30.2 30.7	42.0	45.6 47.6	50.7	88.8 8 4	88.6 88.0	38.2 30.5	42.0	45.6 47.6	50.7	88.8 8 4	88.6 88.0
1953	DB	39.2	42.6	48.0	50.3	88.1	87.9	39.2	42.6	48.0	50.3	88.1	87.9	39.2	42.6	48.0	50.3	88.1	87.9
1954	$_{\rm PR}$	21.1	22.3	22.5	22.3	22.2	93.3	22.9	35.0	25.6	25.7	38.5	26.9	27.9	29.7	30.5	31.0	31.5	32.1
1955	$_{\rm PR}$	19.8	21.0	20.8	20.6	89.1	25.6	32.0	23.9	24.0	36.3	25.1	25.3	27.0	28.8	29.2	29.7	30.2	30.8
1956	РК	19.4	19.3	19.1	85.0	23.7	23.9	22.1	22.3	34.2	23.3	23.5	37.6	26.9 27 2	27.4	27.8	28.3	28.9	29.5
1957	Р. Р. Р.	16.2	17.6	81.3	22.0	22.1	22.3	20.6	32.1	21.5	21.7	35.4	22.9	25.7	26.1	26.5	27.1	27.6 2 <i>6</i> 4	28.2
1959	PR PR	74.6	18.6	18.7	18.8	19.0	19.1	20.2 18.3	18.3 18.3	31.1	23.2 19.4	19.6	21.5 34.5	24.4 23.2	23.6 23.6	24.0	24.6	25.1	25.7 25.7
1960	\mathbf{PR}	17.1	17.1	17.2	17.3	17.5	17.6	16.8	29.2	17.8	17.9	32.4	19.1	22.0	22.4	22.8	23.4	23.9	24.5
1961	$_{\rm PR}$	15.5	15.6	15.7	15.8	15.9	16.0	27.2	16.1	16.2	30.2	17.4	17.6	20.7	21.2	21.6	22.1	22.7	23.3
1962	PR	14.0	14.1	14.1	14.2	14.3	14.4	14.6	14.6	28.2	15.7	15.9	31.6	19.5	19.9	20.4	20.9	21.5	22.0
1963	нц	12.5	12.5	12.6	12.7	12.7	12.8	13.0	26.1	14.0	14.2	29.3	15.4	18.0	18.7	19.2	19.7	20.2	20.8
1965	РВ	9.5	9.5	1.11	9.6	63.8	12.9	10.8	10.9	25.1	12.0	12.1	28.5	15.8	16.3	16.7	17.2	17.8	18.4
1966	PR.	8.0	8.0	8.0	60.4	11.2	11.7	9.4	23.0	10.3	10.5	26.3	12.0	14.7	15.1	15.5	16.0	16.6	17.5
1967	\mathbf{PR}	6.5	6.5	57.1	9.6	10.0	10.2	21.1	8.8	8.9	24.2	10.3	10.5	13.5	13.9	14.4	14.9	15.7	16.3
1968	$_{\rm PR}$	5.1	53.9	8.0	8.4	8.5	8.6	7.3	7.4	22.1	8.6	8.8	25.6	12.4	12.8	13.2	13.9	14.4	15.0
1969	PR	50.9	9.9	6.7	6.9	7.1	7.2	6.0	20.1	7.0	7.2	23.4	8.4	11.3	11.7	12.2	12.8	13.3	13.9
1970	н К	5.2	0.7	0.0 0.0	5.5	5.6 •	5.7	18.2	0.0 •	5.6	21.3	8.0 X 0	6.9	10.2	10.6	11.1	11.6	12.2	12.7
1771		ю и n o	n o n o	5.0 9.0	0.4	4.1	4. o	4.0	4.1	19.0	0.7	0.3	7.7.7	0.0	4.0	ກ່າ	10.4	10.9	11.4
1973		0 0 0 0	0 0 0 0	0 0	3.4	0.0	0 10 0 00	0.0 16.8	4.4	4.5	4.9 19.7	5 2 2 2	0.0	0.0 0.0	* 0 * 0	0.0	2 O T	0.01	11.2
1974	DC	3.0	3.0	3.0	3.1	3.1	60.1	4.0	4.1	18.5	5.1	5.2	21.6	6.8	9.3	9.7	10.1	10.6	11.2
1975	DC	2.7	2.7	2.8	2.8	56.9	5.9	3.8	17.4	4.7	4.8	20.4	5.9	8.8	9.2	9.6	10.1	10.6	11.1
1976	DC	2.5	2.5	2.5	53.8	5.4	5.5	16.4	4.3	4.4	19.2	5.4	5.5	8.8	9.2	9.6	10.0	10.5	11.0
1977	DC	5.5	5.5	50.8	0.0	5.1	0.5 0.7	4.0	4.1	18.1	0.0	5.1	21.1	N 1	9.1	0.2 0.2	6.6 0	10.4	10.9
1978	ט נ ק	75.0	48.0	4.7	4.7 7 7	4, <i>4</i> хі п	4.9 A B	3.8 16 -	17.0 1 3	4.7	4.7	۲9.9 ۲	о т х ∠	2 0 2 0	0.0	9.9 4.0	ກັດ ກັດ	10.3	10.8
1980	DC	4.0	4.1	4.1	4.2	4.2	4.3	4.0	4.1	17.7	5.0	5.1	20.6	8.6	0.8 6.8	9.3	9.7	10.2	10.6
1981	DC	3.8	3.8	3.9	3.9	3.9	4.0	3.8	16.7	4.6	4.7	19.4	5.7	8.6	8.9	9.3	9.7	10.1	10.6
1982	DC	3.6	3.6	3.6	3.6	3.7	3.7	15.7	4.3	4.4	18.3	5.3	5.4	80 10	8.9	9.2	9.6	10.1	10.5
1983	ט מ מ	3.4 1-6	ກຸ ກຸ		 1	3.4 1	ກີ ເ ບີ	4.1 8	4.1 16.3	17.3	4.9	5.0 19.0	20.1 5.6	х х 2	xo xo xo xo	9.7 - 7	0 0 2	10.0	10.4
1985	DO	2.9	2.9	2.8	10.5	52.1	5.6	15.5	4.3	4.4	18.0	5.2	5.3	. 8 4.8	8.7	9.1	9.5 2.6	0.0	10.3
1986	DC	2.7	2.6	2.6	49.3	5.2	5.3	4.1	4.1	17.0	4.9	5.0	19.7	8.4	8.7	9.0	9.4	9.8	10.2
1987	DC	2.4	2.4	46.6	4.9	4.9	5.0	3.8	16.0	4.6	4.6	18.6	5.5	8.3	8.6	9.0	9.3	9.7	10.1
1988		5 15 1 15	44.1	4.6	4.6	4.7	4.7	15.2	4.3	4.3	17.6	5.2	5.2	00 0 00 0	9.0 9.0	6.0 8	6.0 0.0	9.7	10.0
1990		4.1	4.0 1.1	4.1	4.4	4.4	4.4	1.4	4.1 15.7	4.6	4.9	4.9 18.2	5.4	0 0 0 0	o x	n oc o oc	2.0	0 G G	ກ່ອ ກ່ອ
1991	DC	3.9	3.9	3.9	3.9	3.8	3.8	14.9	4.3	4.3	17.2	5.1	5.1	8.2	8.5	80.00	9.1	9.4	9.7
1992	DC	3.7	3.6	3.6	3.6	3.6	3.5	4.1	4.1	16.3	4.8	4.8	18.8	8.2	8.5	8.7	9.0	9.3	9.6
1993	DC	3.5	3.4	3.4	3.3	3.3	3.2	3.9	15.5	4.5	4.5	17.8	5.3	8.1	8.4	8.6	8.9	9.2	9.5
1994	DC	3.2	3.2	3.1	3.1	3.0	50.5	14.7	4.3	4.3	16.9	5.0	5.0	8.1	8 0,0	8.0 1	so i	9.1	9.4
1995		0.0 0	0,1	6.7 0	00 7 01 1	47.9	0.5 7	4.1	4.1	16.0	4.7	4.7	18.3	0 0	n o n	ю. v	8.7 0	0.0	6 - 7
1997		0 9	- 1 - 1 - 1	43.0	40.4 4.6	4.4 9	4.9 4.6	0.0 1 4 4	1.01	4.4	16.5 16.5	1.1.4 4.8	1.0	0.0	0 00 7 -	ю х 4 с	o x	i a a	- 0 0
1998	DO	2.4	40.7	4.4	4.4	4.3	4.3	4.0	4.0	15.6	4.6	4.6	17.8	7.8	8.0	0.0 8.0	6.8	8.7	6.8
1999	DC	38.6	4.2	4.1	4.1	4.1	4.0	3.8	14.8	4.3	4.3	16.9	5.0	7.8	7.9	8.1	8.4	8.6	8.8
2000	DC	4.0	4.0	3.9	3.9	3.9	3.8	14.1	4.2	4.1	16.1	4.7	4.7	7.7	7.9	8.1	8.3	8.5	8.7

TAX	- MALE BI	LUE C.		Scenar.	io B95					Scenari	o B07					Scenari	io AB		
cohor	t/scheme	60	61	retirem 62	tent age 63	64	65	60	61	retiremé 62	ant age 63	64	65	60	61	retireme 62	ent age 63	64	65
1945	DB	57.5	103.9	109.7	110.0	110.3	110.5	57.5	103.9	109.7	110.0	110.3	110.5	57.5	103.9	109.7	110.0	110.3	110.5
1946	DB	58.0	109.3	109.6	109.9	110.2	110.5	58.0	109.3	109.6	109.9	110.2	110.5	58.0	109.3	109.6	109.9	110.2	110.5
1947	DB	63.1	109.3	109.6	109.8	110.1	112.0	63.1	109.3	109.6	109.8	110.1	112.0	63.1	109.3	109.6	109.8	110.1	112.0
1948 1949	a c	62.7	109.2	109.4	111 6	8.111	112.0	62.7	109.2	0.601 109.4	111 6	8.111	112.0	62.7	109.2	0.901 109.4	111 6	8.111	112.0
1950	DB	62.0	109.1	111.4	111.6	111.8	112.0	62.0	109.1	111.4	111.6	111.8	112.0	62.0	109.1	111.4	111.6	111.8	112.0
1951	DB	61.6	111.3	111.5	111.6	111.8	112.0	61.6	111.3	111.5	111.6	111.8	112.0	61.6	111.3	111.5	111.6	111.8	112.0
1952	DB	63.6	111.3	111.5	111.7	111.9	112.0	63.6	111.3	111.5	111.7	111.9	112.0	63.6	111.3	111.5	111.7	111.9	112.0
1953	DB GC	63.5	111.4	111.5	111.7	111.9	112.1	63.5	111.4	111.5	7.111	0,111	112.1	63.5	111.4	111.5	111.7	111.9	112.1
1954	DB	63.1 63.1	111.4	0.111	111.7	111.9	112.1	63.3 63.1	111.4	6.111 111.6	7.111	111.9	112.1	63.1 63.1	111.4 111.4	6.111 6.111	7.111	111.9	112.1
1956	DB	62.9	111.4	111.6	111.7	111.9	112.1	62.9	111.4	111.6	111.7	111.9	112.1	62.9	111.4	111.6	111.7	111.9	112.1
1957	$_{\rm PR}$	31.2	31.1	106.1	36.2	36.6	37.1	34.1	47.7	35.5	35.9	52.5	37.7	39.5	40.2	41.0	41.8	42.7	43.6
1958	PR	29.1	101.2	33.8	34.2	34.6	35.0	45.0	33.2	33.5	49.5	35.2	35.6	37.8	38.5	39.3	40.1	41.0	41.9
1959	ЯЧ	96.6	31.4	31.7	32.1	32.5	32.9	30.8	31.1	46.5	32.7	33.1	51.6	36.1	36.8	37.6	38.4	39.3	40.2
1961	д Н Н Н	29.1	29.4	29.7	3U.U 27 q	30.4 28.2	30.7 28.6	28.8 40.8	43.0 27 g	30.3 28.2	30.0 45.3	48.4 29.9	32.4 30.3	34.4 32.7	33.4	30.9 34 1	35.0 35.0	35.0 35.0	36.8 2
1962	PR.	25.0	25.2	25.5	25.8	26.1	26.4	25.5	25.8	42.3	27.4	27.8	47.5	31.0	31.7	32.4	33.3	34.1	35.1
1963	$_{\rm PR}$	22.9	23.1	23.4	23.7	24.0	24.3	23.5	39.4	24.9	25.3	44.3	27.1	29.2	29.9	30.7	31.5	32.4	33.3
1964	$_{\rm PR}$	20.8	21.1	21.3	21.5	21.8	92.5	36.5	22.6	22.9	41.1	24.6	25.0	27.5	28.2	29.0	29.8	30.7	31.6
1965	PR	18.8	18.9	19.2	19.4	87.4	23.6	20.2	20.5	38.0	22.1	22.5	43.3	25.8	26.5	27.2	28.1	28.9	29.9
1966	PR uc	16.7	16.8	17.0	82.5	21.0	21.4	18.2	35.0	19.6	20.0	40.0	21.8	24.0	24.7	25.5	26.3	27.2	28.1
1069	лч пп	14.6 19 E	7.9.1	7.77	18.6	18.9	19.3	32.1	15.3	17.6	36.9	19.3	19.7	22.3	23.0	23.5	24.6	25.5	26.4
1969	РВ	6.21	13.8	1.01	14.4	14.7	15.1	12.9	30.7	14.4	14.7	35.7	39.U 16.5	18.9	21.3	22.1	21.2	23.0	22.0
1970	PR	11.5	11.7	12.0	12.3	12.6	12.9	27.8	12.0	12.3	32.5	14.0	14.4	17.2	17.9	18.6	19.4	20.3	21.2
1971	$_{\rm PR}$	9.5	9.7	9.9	10.2	10.5	10.8	9.7	10.0	29.4	11.5	11.9	34.5	15.5	16.2	16.9	17.7	18.6	19.5
1972	PR	7.6	7.8	8.0	8.3	8.5	8.7	7.9	26.5	9.3	9.6	31.4	11.3	14.0	14.6	15.4	16.1	17.0	17.9
1973	PR DC	20.00 20.00	5.9	6.1	6.3	6.5	6.7	23.7	7.2	7.4	28.3	9.0	9.3	12.5	13.1	13.8	14.6	15.4	16.2
1974	ט מ ק ב	ი. ი ი	4.0 3.7	4.1 3.8	4.8 9.9	4.4	84.0 7.8	5.1 4.8	5.2 23.2	24.9 6.0	6.3 6.3	6.9 27.9	29.9	10.7	11.4	12.1	12.8	13.5 13.5	14.4 14.3
1976	DC	3.3	3.3	3.4	73.7	7.1	7.4	21.6	5.5	5.7	26.0	7.1	7.4	10.7	11.2	11.9	12.6	13.4	14.2
1977	DC	3.0	3.0	69.1	6.5	6.8	7.0	5.0	5.2	24.3	6.5	6.8	29.1	10.6	11.2	11.8	12.5	13.3	14.0
1978	DC	2.7	64.7	6.0	6.2	6.4	6.6	4.8	22.7	6.0	6.2	27.2	7.6	10.5	11.1	11.7	12.4	13.2	13.9
1979	DC	60.6	5.5	5.6	5.9	6.0	6.2	21.2	5.5	5.6	25.4	7.0	7.3	10.5	11.0	11.6	12.3	13.1	13.8
1980		5.0 7 0	2.5 7.5	0 u 0 0	0.0 0.0	5.7	ດ. ມ	5.0	5.2	23.7	6.4 6 1	6.7 7 <i>6 6</i>	28.4 1 1	10.4	11.0	11.6	12.2	13.0	13.7
1982	D D	4.5	4.6	4.7	4.8	5.0	5.1	20.7	5.4	5.6	24.8	6.9	7.1	10.3	10.8	11.4	12.1	12.8	13.5
1983	DC	4.2	4.3	4.4	4.5	4.6	4.7	5.0	5.2	23.2	6.4	6.6	27.7	10.3	10.8	11.3	12.0	12.7	13.4
1984	DC	3.9	4.0	4.1	4.2	4.2	76.6	4.8	21.7	5.8	6.0	25.9	7.4	10.2	10.7	11.3	11.9	12.6	13.3
1985		0. 1 7 . 0	0.1 7	20 H	50 E	71.9 6.6	7.3 0	20.3	5.4	5.5	24.3	20 U	0.7	10.2	10.7	11.2	8.11	12.5	13.2
1987	DC	3.1	3.1	63.3	0.70 6.3	6.4	0.9	4.8	21.2	5.8	6.0 9	25.4	7.2	10.1	10.5	1111	11.7	12.3	13.0
1988	DC	2.9	59.3	5.8	5.9	6.1	6.2	19.9	5.4	5.5	23.8	6.7	6.9	10.0	10.5	11.0	11.6	12.2	12.9
1989	DC	55.6	5.4	5.5	5.6	5.8	5.9	5.0	5.1	22.2	6.2	6.4	26.5	10.0	10.4	11.0	11.5	12.1	12.8
1990	DC	5.0	5.1	5.2	5.3	5.4	5.5	4.8	20.8	5.8	5.9	24.8	7.1	9.9	10.4	10.9	11.5	12.1	12.7
1991		4.8 1	4.8	4.9	2.0 7 0	5.1	5.5 • 5	19.5	5.4 -	5.5 5.5	23.3	6.6 2	6.8 0 9 0	6.6 0	10.3	10.8	11.4	12.0	12.6
1992		4.0	4.0	4.0	4.1	4 4 0 12	4.0 7.1	0.0	70 A	0 L Z	4 0 7 0	0.0 04.4	0.02	0 x	0.01	0.01 10.7	0.11 11 3	11 8 11 8	19.4
1994	DC	4.0	4.0	4.1	4.1	4.1	70.8	19.2	5.4	5.4	22.9	6.5	6.7	9.7	10.2	10.7	11.2	11.8	12.4
1995	DC	3.8	3.8	3.8	3.8	66.5	6.9	5.0	5.1	21.4	6.1	6.2	25.5	9.7	10.1	10.6	11.1	11.7	12.3
1996	DC	3.5	3.5	3.5	62.5	6.5	6.6	4.8	20.1	5.7	5.8	24.0	6.9	9.7	10.1	10.6	11.1	11.6	12.2
1997		ი. ი.	3.2 0.2	58.6 1	6.0	6.2	6.3 6.3	18.9	ю. 6	5.4	22.5	6.4	6.6	9.6 0	10.1	10.5	11.0	11.6	12.1
1000		3.0 71.6	00.0 7 3	0.7 7 7	о ч о ч	о r o r	о ч о ч	0.0	1.0	21.1 7 6	0.0 7	0.T	1.07	0.0 9.0	10.0	10.1	0.11	11.0	12.1
2000	DO	5.0	5.1	5.1	5.2	5, 5, 5 5, 5	5 C	18.6	5.3	5.4	22.1	6.4	0.0 0.2	0.0 2.0	0.01	10.4	10.9	11.4	11.9
	1		1	1	1	1				1		1				1	1	1	

TAX.	- FEMALE	WHIT	Е С	Scena	rio B95					Scenarie	o B07					Scenari	o AB		
				retirem	ent age				H	etireme	nt age					retireme	ent age		
cohor	t/scheme	60	61	62	63	64	65	60	61	62	63	64	65	60	61	62	63	64	65
1945	DB	34.0	38.0	45.8	49.0	91.0	92.0	34.0	38.0	45.8	49.0	91.0	92.0	34.0	38.0	45.8	49.0	91.0	92.0
1946	DB G	34.3	42.2	45.3	48.5	90.8 00 <i>6</i>	91.8	34.3	42.2	45.3	48.5	90.8	91.8	34.3	42.2	45.3	48.5	90.8 00 <i>6</i>	91.8
1947 1948	DB	30.4 37.8	41.1	44.8 44.2	47.4	90.0 92.3	93.2 93.2	30.4 37.8	41.0	44.8 44.2	47.4 47.4	90.0 92.3	93.2 93.2	38.4 37.8	41.0 41.1	44.8 44.2	47.4 47.4	90.0 92.3	93.2 93.2
1949	DB	37.3	40.6	43.7	49.0	92.1	93.1	37.3	40.6	43.7	49.0	92.1	93.1	37.3	40.6	43.7	49.0	92.1	93.1
1950	DB	36.8	40.1	45.3	48.7	92.0 01.0	93.0	36.8	40.1	45.3	48.7	92.0	93.0	36.8	40.1	45.3	48.7	92.0 01.0	93.0
1951	au au	36.1 37 8	41.6	44.9 44.6	48.3 47 9	91.9 91.7	92.8 0.7 0	36.1 37 8	41.6	44.9 44.6	48.3 47 q	91.9 91.7	92.8 0.7 0	36.1 37 8	41.6 413	44.9 44.6	48.3 47 a	91.9 91.7	92.8
1953	DB	37.4	40.9	44.2	47.6	91.6	92.6	37.4	40.9	44.2	47.6	91.6	92.6	37.4	40.9	44.2	47.6	91.6	92.6
1954	$_{\rm PR}$	20.9	20.8	20.8	20.6	20.6	98.4	22.9	34.4	24.1	24.3	38.3	25.8	29.9	30.5	31.1	31.9	32.6	33.4
1955	\mathbf{PR}	19.2	19.1	19.0	18.8	93.6	24.4	32.2	22.2	22.4	35.9	23.8	24.1	28.6	29.2	29.9	30.6	31.3	32.1
1956	PR	17.6	17.4	17.3	89.0	22.4	22.6	20.5	20.6	33.6	21.8	22.1	37.6	27.4	28.0	28.6	29.3	30.0	30.8
1957 1958	Р. Р.	15.9	15.7 80.8	84.8 18.6	20.4 18.7	20.6 18.0	20.8	18.8 20.2	31.4	20.0 18 3	20.1 32 s	35.1 10.5	21.5	26.2 25.0	26.8 27.5	27.4 26.1	28.1 26.8	28.8 27.5	29.5 28.5
1959	PR	77.1	16.8	16.9	17.0	17.1	17.2	16.4	16.5	30.5	17.7	17.8	34.3	23.8	24.3	24.9	25.5	26.2	26.9
1960	\mathbf{PR}	15.1	15.2	15.2	15.3	15.4	15.4	14.8	28.3	15.8	16.0	31.9	17.2	22.6	23.1	23.6	24.3	24.9	25.6
1961	$_{\rm PR}$	13.5	13.5	13.6	13.6	13.6	13.6	26.2	14.1	14.2	29.6	15.3	15.4	21.3	21.9	22.4	23.0	23.6	24.3
1962	P.R.	11.9	11.9	11.9	11.9	11.8	11.8	12.5	12.5	27.3	13.5	13.5	31.0	20.1	20.6	21.2	21.8	22.4	23.0
1964	д д д	۲0.3 8 7	7.0T	т и 10.7	10.1 8 4	0.01 8 3	71 G	10.9 23 1	101	101	11.8 26.4	1111	1111	10.0 14 8	18.0	18.4 18.4	0.02 20.0	1.12	21.7
1965	P.R.	7.0	6.9	6.8	6.6	67.9	10.0		8.5	24.2	F.07	9.3	27.7	16.6	7.07	17.5	18.0	18.6	19.2
1966	РК	5.4	5.2	5.1	64.2	8.3	8.2	6.9	22.1	7.7	7.7	25.4	8.6	15.4	15.8	16.3	16.8	17.4	17.9
1967	\mathbf{PR}	3.8	3.6	60.7	6.6	6.5	6.5	20.1	6.1	6.0	23.2	6.9	6.8	14.2	14.6	15.1	15.6	16.1	16.6
1968	$_{\rm PR}$	2.2	57.4	5.0	4.9	4.8	4.7	4.6	4.5	21.0	5.2	5.1	24.4	13.1	13.4	13.9	14.4	14.9	15.4
1969	PR	54.2	3.5	3.4	3.2	3.1	2.9	3.0	19.0	3.7	3.6	22.1	4.3	11.9	12.3	12.7	13.2	13.6	14.1
1970	Р.К.	0.7	8 0 1 0	1.7	1.6	1.4	1.2	17.0	21 C	1 50 1 10	20.0	2.7	2.6	10.7	1.11	11.5	12.0	12.4	12.9
1971	ט נ ק	0.4	0.7	0.0	-0.2 -0.2	-0.4 8 0	-0.0	0.7	0.5 16.3	0.0	1 0	10.7	20.6	9.4 7	0 0 0	10.2	10.6	0.11	0.11 11 7
1973		0.0	-0.5	4. 0- 7. 8	0	0.0- 	-1.6	15.2	2.0T	0.5	18.0	1.1	1.4 0.0	4.0 7.0	- 1- 0	1.01	10.5	10.9	11.3
1974	DC	-0.6	-0.9	-1.2	-1.4	-1.7	65.1	0.5	0.3	16.8	0.9	0.7	19.7	6.9	9.6	10.0	10.4	10.8	11.2
1975	DC	-1.0	-1.2	-1.5	-1.8	61.4	1.1	0.2	15.7	0.7	0.5	18.4	1.1	9.2	9.6	9.9	10.3	10.7	11.1
1976	DC	-1.3	-1.6	-1.9	57.8	0.8	0.6	14.7	0.5	0.3	17.3	0.8	0.6	9.2	9.5	9.9	10.3	10.6	11.0
1977	DC	-1.6	-2.0	54.3	0.6	0.4	0.2	0.4	0.2	16.2	0.6	0.4	18.9	9.2	9.5	9.8 0	10.2	10.6	11.0
1978		-2.0	51.1	0.2	0.0 -	0.0	0.0 1	1.0	15.1	0.0	0.3 16 6	17.7	0.7	9.1	9.4 7	0.0 0.0	10.1	10.5	10.9
1980		10.2	0.0	-0.3	-0.5	*·0-	-0.7	0.2	0.0	15.5	0.4	0.1	18.2	9.1 1.6	4.6 4.6	1.6	10.01	10.4	10.7
1981	DC	-0.1	-0.3	-0.6	-0.9	-1.2	-1.6	-0.1	14.6	0.3	0.0	17.0	0.4	9.0	9.3	9.6	9.9	10.3	10.6
1982	DC	-0.4	-0.7	-1.0	-1.3	-1.7	-2.1	13.7	0.2	-0.1	16.0	0.3	0.0	9.0	9.3	9.6	9.9	10.2	10.6
1983	DC	-0.7	-1.0	-1.4	-1.7	-2.1	-2.5	0.1	-0.2	15.0	0.1	-0.2	17.4	9.0 0	9.2	9.5 1	8.0 8.0	10.2	10.5
1985		-1.4	-1.4	-2.1	- 2.5	55.5	-0.1 -0.1	13.2	0.0	0.0	15.3	10.0	-0.4	0.0 0.0	9.1 1.6	9.9 4.6	0.9 7.6	10.01	10.4
1986	DC	-1.7	-2.1	-2.5	52.2	-0.1	-0.5	-0.1	-0.4	14.4	-0.1	-0.5	16.7	8.9	9.1	9.4	9.7	10.0	10.3
1987	DC	-2.0	-2.4	49.2	-0.2	-0.6	-0.9	-0.4	13.5	-0.2	-0.5	15.7	-0.3	8.8	9.1	9.3	9.6	9.9	10.2
1988	DC	-2.3	46.3	-0.3	-0.6	-1.0	-1.4	12.7	-0.3	-0.6	14.7	-0.4	-0.7	so so	9.0	9.3	9.6	9.9	10.2
1989	ט ני מ ב	43.5	n 0- - 0-	- 0.6	-1.0	-1.4	x	-0.0 -	-0.6	13.8 1.0	-0.4 8 0-	15.0	16.0 -0.6	v v x x	0.0	2.0 7.0	ດີດ ກີ	x r 5	10.01
1991	DO	-0.7	-1.0	-1.4	-1.7	-2.2	-2.7	12.2	-0.5	-0.8	14.1	-0.7	-1.1	8.7	6.8	9.5	9.4	9.7	10.0
1992	DC	-1.0	-1.3	-1.7	-2.1	-2.6	-3.1	-0.5	-0.8	13.3	-0.7	-1.1	15.4	8.7	8.9	9.1	9.4	9.6	9.9
1993	DC	-1.3	-1.7	-2.1	-2.5	-3.0	-3.6	-0.8	12.5	-0.7	-1.1	14.4	-1.0	8.7	8.9	9.1	9.3	9.6	9.9
1994	DC	-1.6	-2.0	-2.4	-2.9	-3.4	53.6	11.7	-0.7	-1.1	13.6	-1.0	-1.4	8.6	x o x o	9.1	9.3 5	9.5	9.8 0
1995		-1.9	4.1	ci c x 0	ကို	50.6	ς. Γ. Γ.	-0.7	-1.0	12.7	-1.0	-1.4	14.7	9 0 9 0	xo o xo o	0.0	60 0 19 0	0.0 1.0	0 0 0
1997		-2.6	- 3-0	44.8	-1.2	1.1.6	-2.1	-11.3	0.91	0 F F	-1.4	-1.3	* 1 	0 9 0 8	o x o x	0.6	1.0	9.0 7 4	- 1- 0
1998	DC	-2.9	42.2	-1.2	-1.6	-2.0	-2.5	-0.9	-1.3	12.3	-1.3	-1.7	14.1	10 10	8.7	6.8	9.2	9.4	9.6
1999	DC	39.7	-1.1	-1.5	-2.0	-2.4	-3.0	-1.2	11.5	-1.3	-1.7	13.3	-1.8	8.5	8.7	8.9	9.1	9.3	9.6
2000	DC	-1.1	-1.5	-1.9	-2.3	-2.9	-3.4	10.8	-1.2	-1.6	12.5	-1.7	-2.2	8.5	8.7	8.9	9.1	9.3	9.5

- XAT	· FEMALE	BLUE C		Scenar	io B95					Scenar	io B07					Scenar	·io AB		
cohort	:/scheme	60	61	retirem 62	ent age 63	64	65	60	61	retirem 62	ent age 63	64	65	60	61	retirem 62	ent age 63	64	65
1945	DB	64.3	68.7	121.3	121.4	121.5	121.6	64.3	68.7	121.3	121.4	121.5	121.6	64.3	68.7	121.3	121.4	121.5	121.6
1946	DB	64.9	73.9	121.4	121.5	121.5	121.6	64.9	73.9	121.4	121.5	121.5	121.6	64.9	73.9	121.4	121.5	121.5	121.6
1947	DB	70.1	73.6	121.4	121.4	121.5	123.5	70.1	73.6	121.4	121.4	121.5	123.5	70.1	73.6	121.4	121.4	121.5	123.5
1948	DB CC	69.8	73.3	121.4	121.5	123.6	123.6	69.8	73.3	121.4	121.5	123.6	123.6	69.8	73.3	121.4	121.5	123.6	123.6
1950 1950	90 10 10	69.1 69.1	7.27	123.8	123.8	123.8	123.8	69.1	72.7	123.8	123.8	123.8	123.8	69.1	7.2.7	123.8	123.8	123.8	123.8
1951	DB	68.7	74.9	123.9	123.9	123.9	123.9	68.7	74.9	123.9	123.9	123.9	123.9	68.7	74.9	123.9	123.9	123.9	123.9
1952	DB	71.0	74.8	124.0	124.0	124.0	124.0	71.0	74.8	124.0	124.0	124.0	124.0	71.0	74.8	124.0	124.0	124.0	124.0
1953	DB	70.8	74.7	124.1	124.1	124.1	124.1	70.8	74.7	124.1	124.1	124.1	124.1	70.8	74.7	124.1	124.1	124.1	124.1
1954	DB	70.7	74.6	124.1	124.1	124.1	124.1	70.7	74.6	124.1	124.1	124.1	124.1	70.7	74.6	124.1	124.1	124.1	124.1
1955 1956	D B D B	70.6 33.0	74.4 33.6	124.2 33 3	124.2	124.2 38.6	124.2 38 8	37.0	37.0	124.2 59.6	38.0	124.2 38 3	124.2 к7 к	70.6 77.6	74.4 15.1	124.2	124.2 46.4	124.2	124.2
1957	PR	31.4	31.1	115.1	36.1	36.2	36.3	34.6	49.6	35.6	35.7	54.2	37.0	42.6	43.2	43.8	44.5	45.3	46.1
1958	$_{\rm PR}$	28.9	109.7	33.5	33.6	33.7	33.8	46.6	33.1	33.2	51.0	34.4	34.5	40.7	41.3	41.9	42.6	43.4	44.2
1959	$_{\rm PR}$	104.6	31.0	31.1	31.1	31.2	31.3	30.7	30.7	47.8	31.8	31.9	52.8	38.7	39.3	40.0	40.7	41.5	42.3
1960	PR	28.6	28.6	28.6	28.7	28.7	28.7	28.3	44.7	29.3	29.4	49.4	30.7	36.8	37.4	38.0	38.8	39.5	40.4
1961	P.R.	26.2	26.1	26.1	26.2	26.2	26.1	41.7	26.8	26.8	46.1	28.0	28.2	34.9	35.4	36.1	36.8	37.6	38.5
1962 1963	н Ч	23.7	23.7	23.6	23.6	23.6	23.6	24.3 91 0	24.3 30.7	42.9 22 a	25.5 23.0	25.5 44.6	48.0 24.3	32.9	33.5 31.6	34.2	34.9	35.7 33 8	36.5 34.6
1964	PR	18.8	18.7	18.6	18.5	18.4	100.8	36.7	20.4	20.4	41.3	21.6	21.7	29.0	29.6 29.6	30.3	31.1	31.8	32.7
1965	$_{\rm PR}$	16.3	16.2	16.0	15.9	95.3	19.9	17.9	17.9	37.9	19.0	19.1	43.2	27.0	27.7	28.4	29.1	29.9	30.8
1966	$_{\rm PR}$	13.8	13.6	13.5	89.8	17.3	17.4	15.5	34.7	16.5	16.5	39.7	17.7	25.1	25.7	26.4	27.2	28.0	28.8
1967	$_{\rm PR}$	11.3	1.1.1	84.5	14.7	14.8	14.8	31.6	13.9	14.0	36.3	15.1	15.2	23.1	23.8	24.5	25.2	26.0	26.9
1968	PR	8.7	79.4	12.1	12.2	12.2	12.1	11.5	11.5	32.9	12.6	12.6	38.2	21.2	21.8	22.5	23.3	24.1	24.9
1969	РК	74.4	9.6	9.6	9.6 4	9.6 4	9.0 9	9.0 76.4	29.6 7 E	10.0	10.0	34.7 e e	11.2 8 6	19.2	17.0	20.6	21.3	22.1	23.0
1261	PR	- 4	4.7	4.6	4.5	4.3	4.2	2.1	2.0	27.9	6.1	6.0	33.2	15.4	16.1	16.7	17.5	18.3	19.1
1972	PR	2.4	2.2	2.1	1.9	1.7	1.5	2.8	24.7	3.6	3.5	29.6	4.6	13.6	14.2	14.9	15.6	16.3	1.7.1
1973	DC	0.0	-0.3	-0.5	-0.7	-1.0	-1.3	21.1	1.2	1.1	25.6	2.0	1.8	11.5	12.1	12.7	13.4	14.1	14.9
1974	DC	-0.4	-0.7	-1.0	-1.2	-1.5	93.7	0.9	0.8	23.7	1.6	1.4	28.6	11.5	12.0	12.7	13.3	14.0	14.7
1975	DC	-0.8	-1.1	-1.4	-1.7	87.4	1.9	0.5	21.9	1.2	1.1	26.5	1.9	11.4	12.0	12.6	13.2	13.9	14.6
1976	DC	ς, 1 ε, 1	-1.6	-1.9	81.5	1.6	1.4	20.3	1.0	0.8	24.6	1.6	1.4	11.4	11.9	12.5	13.1	13.8	14.5
1978		-1.7	0.2-	0.07	7 8 - C	0 H	0.0		0.0	0.0	7.7	он 1.1 2.1	4. / Z 1 R	5.11 2.11	0.11 8 11	12.4	13.0	13./ 13.6	14.3
1979	DO	66.0	0.7	0.5	0.3	0.0	-0.3	19.6	0.7	0.5	23.7	1.2	6.0	11.2	11.7	12.3	12.8	13.5	14.1
1980	DC	0.5	0.3	0.0	-0.2	-0.5	-0.9	0.5	0.3	21.9	0.9	0.6	26.4	11.2	11.7	12.2	12.8	13.4	14.0
1981	DC	0.2	-0.1	-0.4	-0.7	-1.1	-1.5	0.2	20.4	0.6	0.4	24.5	1.0	11.1	11.6	12.1	12.7	13.3	13.9
1982	DC	-0.2	-0.5	-0.9	-1.2	-1.6	-2.0	18.9	0.5	0.2	22.8	0.8	0.5	11.1	11.5	12.0	12.6	13.2	13.8
1984		-1.0	-1.0	0 	-1.7	-2.7	84.8	-0.1	19.6	1.12	0.1	23.6	4.0.7 0.5	11.0	11.4	11.9	12.4	13.0	13.6
1985	DC	-1.4	-1.8	-2.2	-2.7	79.2	0.4	18.3	0.2	-0.1	21.9	0.3	0.0	10.9	11.4	11.8	12.4	12.9	13.5
1986	DC	-1.8	-2.2	-2.7	73.9	0.2	-0.2	0.1	-0.2	20.3	0.2	-0.2	24.4	10.9	11.3	11.8	12.3	12.8	13.4
1987		-7.7 -7.7	-2.7	68.9	0.1	-0.3	-0.4	- 0.3	18.9	0.0	-0.3	22.7	0.1	10.8	11.2	11.7	12.2	12.7	13.3
1989		59.9	-0.2	1.0-	4.0- -0.9	0 C 	0 K	0.11	1.0-	-0.4	-0.2	1.0-	-0.0	2.01	1111	0.11	1.21	12.5	13.1
1990	DC	-0.2	-0.6	-1.0	-1.4	-1.8	-2.4	-0.5	18.2	-0.3	-0.7	21.8	-0.4	10.7	11.1	11.5	12.0	12.5	13.0
1991	DC	-0.6	-1.0	-1.4	-1.8	-2.4	-2.9	16.9	-0.4	-0.7	20.2	-0.5	-1.0	10.7	11.0	11.5	11.9	12.4	12.9
1992	DC	-1.0	-1.4	-1.9	-2.3	-2.9	-3.5	-0.4	-0.8	18.8	-0.6	-1.0	22.5	10.6	11.0	11.4	11.9	12.3	12.8
1993	DC	-1.4	-1.8	-2.3	-1- -1-0	-3.4	-4.1	-0.8	17.5	-0.6	-1.0	20.9	-0.9	10.6	10.9	11.3	11.8	12.2	12.7
1995		-2.1	- 2.7	- 2.5	0 0 0 0	- 5.9	-1.2	-0.7	- 1.1	1.1.1	-1.0	0. %	21.6	10.5	10.8	11.2	2.11	12.1	12.6
1996	DC	-2.5	-3.1	-3.7	67.5	-1.2	-1.8	-1.0	16.9	-1.0	-1.4	20.1	-1.4	10.5	10.8	11.2	11.6	12.0	12.5
1997	DC	-2.9	-3.5	62.9	-1.2	-1.7	-2.3	15.7	-1.0	-1.4	18.8	-1.4	-2.0	10.4	10.8	11.1	11.5	12.0	12.4
1998	DC	-3.3	58.7	-1.2	-1.7	-2.3	-2.9	-0.9	-1.4	17.5	-1.4	-1.9	20.8	10.4	10.7	11.1	11.5	11.9	12.3
1999	D C C	54.8	-1.2	-1.7	-2.2	-2.8	-3.4	-1.3	16.3 1 2	-1.4	-1.8	19.4	-1.9	10.4	10.7	11.0	11.4	11.8	12.3
2000	DC	-1.1	0.1-	-2.1	-2.1	-3.3	-4.0	15.1	e.1-	×٦-	18.U	-T.Y	0.2-	10.3	10.6	0.11	11.4	£11.8	12.2

Our papers can be downloaded at:

http://cerp.unito.it/index.php/en/publications

CeRP Working Paper Series

N° 125/11	Agnese Romiti Mariacristina Rossi	Should we Retire Earlier in order to Look After our Parents? The Role of immigrants
N° 124/11	Agnese Romiti	Immigrants-Natives Complementarities in Production: Evidence from Italy
N° 123/11	Elsa Fornero Maria Cristina Rossi Maria Cesira Urzì Brancati	Explaining why, right or wrong, (Italian) households do not like reverse mortgages
N° 122/11	Serena Trucchi	How credit markets affect homeownership: an explanation based on differences between Italian regions
N° 121/11	Elsa Fornero Chiara Monticone Serena Trucchi	The effect of financial literacy on mortgage choices
N° 120/11	Giovanni Mastrobuoni Filippo Taddei	Age Before Beauty? Productivity and Work vs. Seniority and Early Retirement
N° 119/11	Maarten van Rooij Annamaria Lusardi Rob Alessie	Financial Literacy, Retirement Planning, and Household Wealth
N° 118/11	Luca Beltrametti Matteo Della Valle	Does the implicit pension debt mean anything after all?
N° 117/11	Riccardo Calcagno Chiara Monticone	Financial Literacy and the Demand for Financial Advice
N° 116/11	Annamaria Lusardi Daniel Schneider Peter Tufano	Financially Fragile Households: Evidence and Implications
N° 115/11	Adele Atkinson Flore-Anne Messy	Assessing financial literacy in 12 countries: an OECD Pilot Exercise
N° 114/11	Leora Klapper Georgios A. Panos	Financial Literacy and Retirement Planning in View of a Growing Youth Demographic: The Russian Case
N° 113/11	Diana Crossan David Feslier Roger Hurnard	Financial Literacy and Retirement Planning in New Zealand
N° 112/11	Johan Almenberg Jenny Säve-Söderbergh	Financial Literacy and Retirement Planning in Sweden
N° 111/11	Elsa Fornero Chiara Monticone	Financial Literacy and Pension Plan Participation in Italy
N° 110/11	Rob Alessie Maarten Van Rooij Annamaria Lusardi	Financial Literacy, Retirement Preparation and Pension Expectations in the Netherlands
N° 109/11	Tabea Bucher-Koenen Annamaria Lusardi	Financial Literacy and Retirement Planning in Germany
N° 108/11	Shizuka Sekita	Financial Literacy and Retirement Planning in Japan
N° 107/11	Annamaria Lusardi Olivia S. Mitchell	Financial Literacy and Retirement Planning in the United States

N° 106/11	Annamaria Lusardi Olivia S. Mitchell	Financial Literacy Around the World: An Overview
N° 105/11	Agnese Romiti	Immigrants-natives complementarities in production: evidence from Italy
N° 104/11	Ambrogio Rinaldi	Pension awareness and nation-wide auto-enrolment: the Italian experience
N° 103/10	Fabio Bagliano Claudio Morana	The Great Recession: US dynamics and spillovers to the world economy
N° 102/10	Nuno Cassola Claudio Morana	The 2007-? financial crisis: a money market perspective
N° 101/10	Tetyana Dubovyk	Macroeconomic Aspects of Italian Pension Reforms of 1990s
N° 100/10	Laura Piatti Giuseppe Rocco	L'educazione e la comunicazione previdenziale - Il caso italiano
N° 99/10	Fabio Bagliano Claudio Morana	The effects of US economic and financial crises on euro area convergence
N° 98/10	Annamaria Lusardi Daniel Schneider Peter Tufano	The Economic Crisis and Medical Care Usage
N° 97/10	Carlo Maccheroni Tiziana Barugola	E se l'aspettativa di vita continuasse la sua crescita? Alcune ipotesi per le generazioni italiane 1950-2005
N° 96/10	Riccardo Calcagno Mariacristina Rossi	Portfolio Choice and Precautionary Savings
N° 95/10	Flavia Coda Moscarola Elsa Fornero Mariacristina Rossi	Parents/children "deals": Inter-Vivos Transfers and Living Proximity
N° 94/10	John A. List Sally Sadoff Mathis Wagner	So you want to run an experiment, now what? Some Simple Rules of Thumb for Optimal Experimental Design
N° 93/10	Mathis Wagner	The Heterogeneous Labor Market Effects of Immigration
N° 92/10	Rob Alessie Michele Belloni	Retirement choices in Italy: what an option value model tells us
N° 91/09	Annamaria Lusardi Olivia S. Mitchell Vilsa Curto	Financial Literacy among the Young: Evidence and Implications for Consumer Policy
N° 90/09	Annamaria Lusardi Olivia S. Mitchell	How Ordinary Consumers Make Complex Economic Decisions: Financial Literacy and Retirement Readiness
N° 89/09	Elena Vigna	Mean-variance inefficiency of CRRA and CARA utility functions for portfolio selection in defined contribution pension schemes
N° 88/09	Maela Giofré	Convergence of EMU Equity Portfolios
N° 87/09	Elsa Fornero Annamaria Lusardi Chiara Monticone	Adequacy of Saving for Old Age in Europe
N° 86/09	Margherita Borella Flavia Coda Moscarola	Microsimulation of Pension Reforms: Behavioural versus Nonbehavioural Approach
N° 85/09	Cathal O'Donoghue John Lennon	The Life-Cycle Income Analysis Model (LIAM): A Study of a Flexible Dynamic Microsimulation Modelling Computing

	Stephen Hynes	Framework
N° 84/09	Luca Spataro	Il sistema previdenziale italiano dallo shock petrolifero del 1973 al Trattato di Maastricht del 1993
N° 83/09	Annamaria Lusardi Peter Tufano	Debt Literacy, Financial Experiences, and Overindebtedness
N° 82/09	Carolina Fugazza Massimo Guidolin Giovanna Nicodano	Time and Risk Diversification in Real Estate Investments: Assessing the Ex Post Economic Value
N° 81/09	Fabio Bagliano Claudio Morana	Permanent and Transitory Dynamics in House Prices and Consumption: Cross-Country Evidence
N° 80/08	Claudio Campanale	Learning, Ambiguity and Life-Cycle Portfolio Allocation
N° 79/08	Annamaria Lusardi	Increasing the Effectiveness of Financial Education in the Workplace
N° 78/08	Margherita Borella Giovanna Segre	Le pensioni dei lavoratori parasubordinati: prospettive dopo un decennio di gestione separata
N° 77/08	Giovanni Guazzarotti Pietro Tommasino	The Annuity Market in an Evolving Pension System: Lessons from Italy
N° 76/08	Riccardo Calcagno Elsa Fornero Mariacristina Rossi	The Effect of House Prices on Household Saving: The Case of Italy
N° 75/08	Harold Alderman Johannes Hoogeveen Mariacristina Rossi	Preschool Nutrition and Subsequent Schooling Attainment: Longitudinal Evidence from Tanzania
N° 74/08	Maela Giofré	Information Asymmetries and Foreign Equity Portfolios: Households versus Financial Investors
N° 73/08	Michele Belloni Rob Alessie	The Importance of Financial Incentives on Retirement Choices: New Evidence for Italy
N° 72/08	Annamaria Lusardi Olivia Mitchell	Planning and Financial Literacy: How Do Women Fare?
N° 71/07	Flavia Coda Moscarola	Women participation and caring decisions: do different institutional frameworks matter? A comparison between Italy and The Netherlands
N° 70/07	Radha Iyengar Giovanni Mastrobuoni	The Political Economy of the Disability Insurance. Theory and Evidence of Gubernatorial Learning from Social Security Administration Monitoring
N° 69/07	Carolina Fugazza Massimo Guidolin Giovanna Nicodano	Investing in Mixed Asset Portfolios: the Ex-Post Performance
N° 68/07	Massimo Guidolin Giovanna Nicodano	Small Caps in International Diversified Portfolios
N° 67/07	Carolina Fugazza Maela Giofré Giovanna Nicodano	International Diversification and Labor Income Risk
N° 66/07	Maarten van Rooij Annamaria Lusardi Rob Alessie	Financial Literacy and Stock Market Participation

N° 65/07	Annamaria Lusardi	Household Saving Behavior: The Role of Literacy, Information and Financial Education Programs (Updated version June 08: "Financial Literacy: An Essential Tool for Informed Consumer Chains?")
N° 64/07	Carlo Casarosa Luca Spataro	Rate of Growth of Population, Saving and Wealth in the Basic Life-cycle Model when the Household is the Decision Unit
N° 63/07	Claudio Campanale	Life-Cycle Portfolio Choice: The Role of Heterogeneous Under- Diversification
N° 62/07	Margherita Borella Elsa Fornero Mariacristina Rossi	Does Consumption Respond to Predicted Increases in Cash-on- hand Availability? Evidence from the Italian "Severance Pay"
N° 61/07	Irina Kovrova	Effects of the Introduction of a Funded Pillar on the Russian Household Savings: Evidence from the 2002 Pension Reform
N° 60/07	Riccardo Cesari Giuseppe Grande Fabio Panetta	La Previdenza Complementare in Italia: Caratteristiche, Sviluppo e Opportunità per i Lavoratori
N° 59/07	Riccardo Calcagno Roman Kraeussl Chiara Monticone	An Analysis of the Effects of the Severance Pay Reform on Credit to Italian SMEs
N° 58/07	Elisa Luciano Jaap Spreeuw Elena Vigna	Modelling Stochastic Mortality for Dependent Lives
N° 57/07	Giovanni Mastrobuoni Matthew Weinberg	Heterogeneity in Intra-Monthly Consumption. Patterns, Self- Control, and Savings at Retirement
N° 56/07	John A. Turner Satyendra Verma	Why Some Workers Don't Take 401(k) Plan Offers: Inertia versus Economics
N° 55/06	Antonio Abatemarco	On the Measurement of Intra-Generational Lifetime Redistribution in Pension Systems
N° 54/06	Annamaria Lusardi Olivia S. Mitchell	Baby Boomer Retirement Security: The Roles of Planning, Financial Literacy, and Housing Wealth
N° 53/06	Giovanni Mastrobuoni	Labor Supply Effects of the Recent Social Security Benefit Cuts: Empirical Estimates Using Cohort Discontinuities
N° 52/06	Luigi Guiso Tullio Jappelli	Information Acquisition and Portfolio Performance
N° 51/06	Giovanni Mastrobuoni	The Social Security Earnings Test Removal. Money Saved or Money Spent by the Trust Fund?
N° 50/06	Andrea Buffa Chiara Monticone	Do European Pension Reforms Improve the Adequacy of Saving?
N° 49/06	Mariacristina Rossi	Examining the Interaction between Saving and Contributions to Personal Pension Plans. Evidence from the BHPS
N° 48/06	Onorato Castellino Elsa Fornero	Public Policy and the Transition to Private Pension Provision in the United States and Europe
N° 47/06	Michele Belloni Carlo Maccheroni	Actuarial fairness when longevity increases: an evaluation of the Italian pension system
N° 46/05	Annamaria Lusardi Olivia S. Mitchell	Financial Literacy and Planning: Implications for Retirement Wellbeing
N° 45/05	Claudio Campanale	Increasing Returns to Savings and Wealth Inequality
N° 44/05	Henrik Cronqvist	Advertising and Portfolio Choice

N° 43/05	John Beshears James J. Choi David Laibson	The Importance of Default Options for Retirement Saving Outcomes: Evidence from the United States
N° 42/05	Brigitte C. Madrian Margherita Borella Flavia Coda Moscarola	Distributive Properties of Pensions Systems: a Simulation of the Italian Transition from Defined Benefit to Defined Contribution
N° 41/05	Massimo Guidolin Giovanna Nicodano	Small Caps in International Equity Portfolios: The Effects of Variance Risk.
N° 40/05	Carolina Fugazza Massimo Guidolin Giovanna Nicodano	Investing for the Long-Run in European Real Estate. Does Predictability Matter?
N° 39/05	Anna Rita Bacinello	Modelling the Surrender Conditions in Equity-Linked Life Insurance
N° 38/05	Carolina Fugazza Federica Teppa	An Empirical Assessment of the Italian Severance Payment (TFR)
N° 37/04	Jay Ginn	Actuarial Fairness or Social Justice? A Gender Perspective on Redistribution in Pension Systems
N° 36/04	Laurence J. Kotlikoff	Pensions Systems and the Intergenerational Distribution of Resources
N° 35/04	Monika Bütler Olivia Huguenin Federica Teppa	What Triggers Early Retirement. Results from Swiss Pension Funds
N° 34/04	Chourouk Houssi	Le Vieillissement Démographique : Problématique des Régimes de Pension en Tunisie
N° 33/04	Elsa Fornero Carolina Fugazza Giacomo Ponzetto	A Comparative Analysis of the Costs of Italian Individual Pension Plans
N° 32/04	Angelo Marano Paolo Sestito	Older Workers and Pensioners: the Challenge of Ageing on the Italian Public Pension System and Labour Market
N° 31/03	Giacomo Ponzetto	Risk Aversion and the Utility of Annuities
N° 30/03	Bas Arts Elena Vigna	A Switch Criterion for Defined Contribution Pension Schemes
N° 29/02	Marco Taboga	The Realized Equity Premium has been Higher than Expected: Further Evidence
N° 28/02	Luca Spataro	New Tools in Micromodeling Retirement Decisions: Overview and Applications to the Italian Case
N° 27/02	Reinhold Schnabel	Annuities in Germany before and after the Pension Reform of 2001
N° 26/02	E. Philip Davis	Issues in the Regulation of Annuities Markets
N° 25/02	Edmund Cannon Ian Tonks	The Behaviour of UK Annuity Prices from 1972 to the Present
N° 24/02	Laura Ballotta Steven Haberman	Valuation of Guaranteed Annuity Conversion Options
N° 23/02	Ermanno Pitacco	Longevity Risk in Living Benefits
N° 22/02	Chris Soares Mark Warshawsky	Annuity Risk: Volatility and Inflation Exposure in Payments from Immediate Life Annuities
N° 21/02	Olivia S. Mitchell David McCarthy	Annuities for an Ageing World

N° 20/02	Mauro Mastrogiacomo	Dual Retirement in Italy and Expectations
N° 19/02	Paolo Battocchio Francesco Menoncin	Optimal Portfolio Strategies with Stochastic Wage Income and Inflation: The Case of a Defined Contribution Pension Plan
N° 18/02	Francesco Daveri	Labor Taxes and Unemployment: a Survey of the Aggregate Evidence
N° 17/02	Richard Disney and Sarah Smith	The Labour Supply Effect of the Abolition of the Earnings Rule for Older Workers in the United Kingdom
N° 16/01	Estelle James and Xue Song	Annuities Markets Around the World: Money's Worth and Risk Intermediation
N° 15/01	Estelle James	How Can China Solve ist Old Age Security Problem? The Interaction Between Pension, SOE and Financial Market Reform
N° 14/01	Thomas H. Noe	Investor Activism and Financial Market Structure
N° 13/01	Michela Scatigna	Institutional Investors, Corporate Governance and Pension Funds
N° 12/01	Roberta Romano	Less is More: Making Shareholder Activism a Valuable Mechanism of Corporate Governance
N° 11/01	Mara Faccio and Ameziane Lasfer	Institutional Shareholders and Corporate Governance: The Case of UK Pension Funds
N° 10/01	Vincenzo Andrietti and Vincent Hildebrand	Pension Portability and Labour Mobility in the United States. New Evidence from the SIPP Data
N° 9/01	Hans Blommestein	Ageing, Pension Reform, and Financial Market Implications in the OECD Area
N° 8/01	Margherita Borella	Social Security Systems and the Distribution of Income: an Application to the Italian Case
N° 7/01	Margherita Borella	The Error Structure of Earnings: an Analysis on Italian Longitudinal Data
N° 6/01	Flavia Coda Moscarola	The Effects of Immigration Inflows on the Sustainability of the Italian Welfare State
N° 5/01	Vincenzo Andrietti	Occupational Pensions and Interfirm Job Mobility in the European Union. Evidence from the ECHP Survey
N° 4/01	Peter Diamond	Towards an Optimal Social Security Design
N° 3/00	Emanuele Baldacci Luca Inglese	Le caratteristiche socio economiche dei pensionati in Italia. Analisi della distribuzione dei redditi da pensione (only available in the Italian version)
N° 2/00	Pier Marco Ferraresi Elsa Fornero	Social Security Transition in Italy: Costs, Distorsions and (some) Possible Correction
N° 1/00	Guido Menzio	Opting Out of Social Security over the Life Cycle